# University of New Hampshire Scholars' Repository

Doctoral Dissertations Student Scholarship

Summer 1972

# ON REDEI SUBGROUPS CONTAINING THEIR CENTRALIZERS

ROBERT EDWARD MCDONALD

Follow this and additional works at: https://scholars.unh.edu/dissertation

### Recommended Citation

MCDONALD, ROBERT EDWARD, "ON REDEI SUBGROUPS CONTAINING THEIR CENTRALIZERS" (1972). Doctoral Dissertations. 995.

https://scholars.unh.edu/dissertation/995

This Dissertation is brought to you for free and open access by the Student Scholarship at University of New Hampshire Scholars' Repository. It has been accepted for inclusion in Doctoral Dissertations by an authorized administrator of University of New Hampshire Scholars' Repository. For more information, please contact nicole.hentz@unh.edu.

### **INFORMATION TO USERS**

This dissertation was produced from a microfilm copy of the original document. While the most advanced technological means to photograph and reproduce this document have been used, the quality is heavily dependent upon the quality of the original submitted.

The following explanation of techniques is provided to help you understand markings or patterns which may appear on this reproduction.

- The sign or "target" for pages apparently lacking from the document photographed is "Missing Page(s)". If it was possible to obtain the missing page(s) or section, they are spliced into the film along with adjacent pages. This may have necessitated cutting thru an image and duplicating adjacent pages to insure you complete continuity.
- When an image on the film is obliterated with a large round black mark, it is an indication that the photographer suspected that the copy may have moved during exposure and thus cause a blurred image. You will find a good image of the page in the adjacent frame.
- 3. When a map, drawing or chart, etc., was part of the material being photographed the photographer followed a definite method in "sectioning" the material. It is customary to begin photoing at the upper left hand corner of a large sheet and to continue photoing from left to right in equal sections with a small overlap. If necessary, sectioning is continued again beginning below the first row and continuing on until complete.
- 4. The majority of users indicate that the textual content is of greatest value, however, a somewhat higher quality reproduction could be made from "photographs" if essential to the understanding of the dissertation. Silver prints of "photographs" may be ordered at additional charge by writing the Order Department, giving the catalog number, title, author and specific pages you wish reproduced.

**University Microfilms** 

300 North Zeeb Road Ann Arbor, Michigan 48106 A Xerox Education Company

73-4355

MCDONALD, Robert Edward, 1941-ON REDEI SUBGROUPS CONTAINING THEIR CENTRALIZERS.

University of New Hampshire, Ph.D., 1972 Mathematics

University Microfilms, A XEROX Company, Ann Arbor, Michigan

# ON REDEI SUBGROUPS CONTAINING THEIR CENTRALIZERS

bу

ROBERT E. MCDONALD
M.A., University of Massachusetts, 1965

### A THESIS

Submitted to the University of New Hampshire

In Partial Fulfillment of

The Requirements for the Degree of

Doctor of Philosophy
Graduate School
Department of Mathematics
August, 1972

### PLEASE NOTE:

Some pages may have indistinct print. Filmed as received.

University Microfilms, A Xerox Education Company

## ACKNOWLEDGEMENT

The author wishes to express his appreciation and gratitude to Professor Homer F. Bechtell for his guidance, assistance, and encouragement during the course of this research.

He also wishes to thank Patricia, his wife, for her patience and understanding.

Filmed as received without page(s) iii .

UNIVERSITY MICROFILMS.

## TABLE OF CONTENTS

	ABSTRACT	V
	INTRODUCTION	1
I.	FUNDAMENTAL CONCEPTS	9
II.	REDEI SUBGROUPS AS CENTRALIZERS OF THEIR CENTERS	19
EII.	REDEI SUBGROUPS WHICH CONTAIN THEIR CENTRALIZERS	34
IV.	ALL SUBGROUPS OF A FIXED INDEX ARE REDEI SUBGROUPS	67
٧.	SUMMARY	80
	BTBLIOGRAPHY	83

#### ABSTRACT

# ON REDEI SUBGROUPS CONTAINING THEIR CENTRALIZERS

by

### ROBERT E. MCDONALD

Three classes of finite p-groups for p > 2 are considered in this thesis, each class being defined by conditions which are placed on the minimal nonabelian subgroups of a group. The non-abelian subgroups for which each proper subgroup is abelian are called Redei groups.

Class  $\mathcal J$  is the collection of nonabelian p-groups for which the centralizer of a Redei subgroup is its center. Class  $\mathcal R$  is the collection of nonabelian p-groups for which each Redei subgroup is the centralizer of its center. A third class  $\mathcal J^*$  is the collection of all nonabelian p-groups for which all subgroups for some given index are Redei. The three classes are shown to satisfy the following inclusion relationships.  $\mathcal R \subset \mathcal J^* \subset \mathcal J$  . All three classes are subgroup inherited.

The two properties on Redei subgroups which define class  $\mathcal Z$  and class  $\mathcal R$ , respectively, are shown to extend to all nonabelian subgroups. For G  $\epsilon$   $\mathcal Z$ ,  $Z(N)=\mathcal C_G(N)$  for each nonabelian subgroup N of G iff  $Z(R)=\mathcal C_G(R)$  for each Redei subgroup R of G .

For  $G \in \mathcal{R}$  ,  $N = C_G(Z(N))$  for each nonabelian subgroup N of G iff  $R = C_G(Z(R))$  for each Redei subgroup R of G . A result of this second equivalence is that for  $G \in \mathcal{R}$  ,  $(Z(N)\colon Z(M)) = (M\colon N)$  for each pair of nonabelian subgroups N , M such that  $N \subseteq M \subseteq G$  .

It is shown that a group in class R is either a metacyclic group or a nonmetacyclic Redei group. This leads to a characterization of non-Redei metacyclic p-groups as groups for which each Redei subgroup is the centralizer of its center.

It is found that class  $3^*$  is "almost equal" to class R in that, for p > 3, there is only one type of group in  $3^*$  that is not in R. These groups have order  $p^5$ . For p = 3, there is also only one type of group in  $3^*$  that is not in R. Each of these groups has its Redei subgroups of order  $p^4$ . As a consequence, a second characterization is obtained for non-Redei metacyclic p-groups as groups for which there is an abelian subgroup of order  $p^4$  and for which each subgroup of index  $p^1$  is Redei if there is one Redei subgroup of that index.

The minimal non-Redei subgroups of class  $\it 3$  are identified. These are precisely the nonabelian groups for which there is at most one abelian maximal subgroup and for which each subgroup of index  $\it p^2$  is abelian.

#### INTRODUCTION

Several means by which a group's structure is controlled by its minimal nonabelian subgroups are investigated for finite nonabelian p-groups for p > 2. The nonabelian p-groups for which each proper subgroup is abelian have been examined by Miller-Moreno [19] and have been completely described in terms of defining relations by Redei [20]. Such nonabelian p-groups are minimal nonabelian subgroups of a group and are called Redei groups throughout this study.

One common approach to the problem of classifying nonabelian groups is to stress the type of abelian subgroups possessed by the groups. While surveying the literature with regard to this approach it has become apparent that attention continually reverts to the existence and the properties of the Redei subgroups. This raises the question as to whether or not Redei subgroups can assume more than passive control over the structure of a group. In this regard, a new approach has been employed whereby the dominant role in examining group structures is assigned to the Redei subgroups. This can be accomplished by placing conditions on the minimal nonabelian subgroups whereas the previous approach concentrated on choosing specific types of abelian subgroups.

The next few examples about well known results illustrate the effectiveness of Redei groups in this previous approach.

Example 1: Burnside [6] classified the nonabelian groups of order p 4 by examining the group structure through the maximal cyclic subgroups of the group. Huppert [15] derived the same classification,

but through the maximal abelian subgroups of the group. In the cases where the group was not itself a Redei group, both authors were required to examine the Redei subgroups of order  $\ p^3$  .

Example 2: The paper of Gaschütz [9] revived interest in the properties of the Frattini subgroup  $\Phi(G)$ , the intersection of the maximal subgroups of a group G. Hobby [14] proved that  $\Phi(G)$  for a nonabelian p-group G cannot have a cyclic center unless  $\Phi(G)$  is itself cyclic. This extended Burnside's result which used the commutator subgroup G' rather than  $\Phi(G)$  in this statement. In order to prove his result it was necessary for Hobby to show that no Redei group of order  $\mathbb{P}^3$  can be the Frattini subgroup of a group.

Example 3: Several significant works dealing with simple groups (e.g., Feit and Thompson [7], Thompson [21], and Gorenstein [10]) continually refer to a theorem of P. Hall [11]. For clarification two definitions are first introduced. For the product G = ER to be a central product, the properties  $E \cap R \subseteq Z(G)$  and  $E \subseteq C_G(R)$  must be satisfied where  $C_G(R)$  is the centralizer in G of R. In addition, a group E is defined as extraspecial if  $E' = \phi(G) = Z(G)$  and the order of E' is P. In his theorem Hall has classified a nonabelian p-group G with no non-cyclic characteristic abelian subgroups as a central product ER where E = 1 or E is extraspecial and R is cyclic or P = 2 and R is isomorphic to the generalized dihedral, the generalized quaternion or the symmetric group of order P0 for P1. To obtain a complete classification of these groups, Gorenstein [10] has elaborated on the class of extraspecial P1. He has proved that an extraspecial P2 group P3 for P4 and P3 for P4 and P4 are complete class of extraspecial P4 for P5 for P6 for P8 and P9 for P9 for

has order  $p^{2r+1}$  and is the central product of  $r \ge 1$  Redei subgroups of order  $p^3$ . Indeed, here are two classes of groups, the extraspecial groups and the groups in Hall's theorem, where the structure of the groups is described in terms of Redei subgroups.

Moreover, the proof of Gorenstein's result is also dependent upon Redei subgroups and their hold over the structure of the group. For this reason an outline of the proof is included. Two results are needed.

- (1)  $P = CC_p(C)$  if C is an extraspecial subgroup of the p-group P such that  $[P,C] \subseteq Z(G)$ . [P,C] is the subgroup of P generated by the commutators  $[p,c] = p^{-1}c^{-1}pc$  for any  $p \in P$  and any  $c \in C$ .
- (2) Each extraspecial group G has a Redei subgroup R of order  $p^3$  for which  $[G,R] \subseteq [G,G] = G' = Z(G)$ .

At this time it should be noted that each Redei group of order  $p^3$  is an extraspecial group. By the application of (1) and (2) to an extraspecial group G,  $G = RC_G(R)$  where R is a Redei subgroup of order  $p^3$ . If  $C_G(R)$  is abelian, then  $C_G(R) \subseteq Z(G) = Z(R)$  so that G = R. If, on the other hand,  $C_G(R)$  is nonabelian, then  $R \subseteq G$  properly. It can be shown that  $C_G(R)$  is also an extraspecial group. So, an inductive argument can now be applied to arrive at the desired result. Although the Redei subgroups of order  $p^3$  control the direction of the proof, it is significant to note that the role of the Redei subgroups is nevertheless passive since their existence follows from the group structure under examination.

Example 4: In line with the approach which stresses the role of abelian subgroups is the problem of whether or not a group possesses a

normal abelian subgroup of a given order when this same group has an abelian subgroup of that order. As a partial answer, Alperin [1] showed that if a group has an abelian subgroup of index  $p^3$  then the group has a normal abelian subgroup of index  $p^3$ . Konvisser [17] generalized the above to show that if a normal subgroup N of a group G has an abelian subgroup of index  $p^2$ , then N has an abelian subgroup of index  $p^2$  that is normal in G.

One aspect of this problem revolves around the structure of a group with an abelian maximal subgroup. This structure is derived from the power imparted to the Redei subgroups by the abelian maximal subgroups. If the group is not itself a Redei group, then there are Redei subgroups properly contained in G . If G has two abelian maximal subgroups, it can be shown that each Redei subgroup is a normal subgroup of G and that G can be written as a central product involving a Redei subgroup. If G has only one abelian maximal subgroup, then G is a product of this abelian maximal subgroup and of any Redei subgroup. This product is not a central product. However, the G-normalizer of the Redei subgroup R has nilpotent class at most three; that is, if N is the normalizer, then the commutators  $a^{-1}b^{-1}ab \in Z(N)$  for  $a \in N$  and  $b \in N'$ .

These four examples are some indication of how Redei subgroups influence a group's structure when the group possesses particular abelian subgroups. In contrast, this thesis introduces a new aspect of the Redei subgroups' hold over the group's structure—namely, through the classification of groups according to properties possessed by their Redei subgroups.

Somewhat natural conditions to be examined for the minimal nonabelian subgroups in this approach are suggested by the maximal abelian subgroups which, in concept, are dual to the Redei subgroups. It is well known [15, p. 302] that  $M = C_{\mathbb{G}}(M)$  for each maximal abelian subgroup M of a group G . Since each abelian subgroup is its own center, the above equality can be written as either  $Z(M) = C_{\mathbb{G}}(M)$  or  $M = C_{\mathbb{G}}(Z(M))$ . These two forms provide the motivation behind two of the properties which are placed on the Redei subgroups in this work:

- (1)  $Z(R) = C_G(R)$  for each Redei subgroup R , and
- (2)  $R = C_{G}(Z(R))$  for each Redei subgroup R .

Altogether, there are three properties placed on the Redei subgroups, each of which defines a class of finite nonabelian p-groups for p > 2. Class g and class g are defined by properties (1) and (2), respectively:

 $\mathcal{J} = \{G \mid Z(R) = C_G(R) \text{ for each Redei subgroup } R \text{ of } G \} \text{ and }$   $R = \{G \mid R = C_G(Z(R)) \text{ for each Redei subgroup } R \text{ of } G \} .$  The third class  $\mathcal{J}^*$  is defined by the property:

(3) If R is a Redei subgroup of G of index  $p^1$ , then each subgroup of index  $p^1$  is a Redei group.

There are five chapters in this thesis. Chapter I contains the prerequisite definitions and fundamental theorems needed throughout this work.

The groups in R are classified in Chapter II. For this purpose a fourth class  $R^*$  is introduced by means of the property:

(4)  $N = C_G(Z(N))$  for each nonabelian subgroup N of G . This property is a stronger version of property (2).

A group is defined as metacyclic if there is a cyclic normal sub-

group whose factor group is also cyclic. One of the main results of Chapter II shows that a nonabelian group G  $\epsilon$  R is either a metacyclic group or a nonmetacyclic Redei group. Another outcome identifies the class of nonabelian metacyclic groups as a subclass of R. Consequences of these two results are equality of class R and R and a new characterization for the class of nonabelian metacyclic groups.

Corresponding to the method of Alperin [1] where he considered large abelian subgroups of a group, the investigation of class 3 in Chapter III considers groups from class 3 with a Redei maximal subgroup. It is first determined that a group from this subclass of 3 is one of two types:

- (i) a group for which each subgroup of index p<sup>2</sup> is abelian, or
- (ii) a group with no abelian maximal subgroup but with a Redei subgroup which is not a maximal subgroup.

It is shown that the groups in (ii) are affected by the groups in (i). Also, the groups in (i) with an abelian maximal subgroup are completely characterized.

The investigation started in Chapter III is continued in Chapter IV. The groups of type (i) for which each maximal subgroup is Redei are described. These are precisely the groups in  $3^*$  for which each subgroup of index  $p^2$  is abelian. It is found that class  $3^*$  is "almost equal" to class  $\ell$ . Furthermore, a second characterization of non-Redei metacyclic p-groups is obtained.

Chapter V contains a summary of the results presented here and identifies unanswered related questions that provide a basis for future investigations with respect to Redei subgroups in p-groups.

### NOTATION

Only finite p-groups for primes p > 2 are considered.

|G| - order of the group G

a ε G - a is an element of G

 $< a_1, a_2, \cdots, a_n > -$  the subgroup generated by the elements  $a_1, a_2, \cdots, a_n$ 

 $A \subseteq B$  - A is a subset of B

 $A \sim B$  - the set of elements from A that are not elements of B

 $A \leq G$  - A is a subgroup of G

A < G - A is a proper subgroup of G

AB - product of A and B; < ab a & A and b & B >

(G:A) - the index of A in G

A ⊲ G - A is a normal subgroup of G

[a,b] - the commutator,  $a^{-1}b^{-1}ab$ , of the elements a and b

[A,B] - < [a,b]|a  $\in$  A, b  $\in$  B >

G' - the commutator subgroup of G , [G,G]

 $G_i$  - the i<sup>th</sup> term of the lower central series where  $G_1 = G$  and  $G_i = [G,G_{i-1}]$ 

class of  $G = c - G_c \neq 1$  but  $G_{c+1} = 1$ 

Z(G) - the center of G

 $\Phi(G)$  - the Frattini subgroup of G; the intersection of all the maximal subgroups of G

 $\Omega_1(G)$  - the subgroup of G generated by the elements of order p; < g|g  $\epsilon$  G and  $g^p$  = 1 >

 $v_1(G)$  - the subgroup of G generated by the p<sup>th</sup> powers of elements of G; <  $g^p | g \in G >$ 

- $C_{G}(N)$  the G-centralizer of N; < g|g  $\epsilon$  G and [g,n] = 1 >
- $N_{G}(N)$  the G-normalizer of N; < g|g  $\epsilon$  G and g<sup>-1</sup>Ng = N >
- type (m,n) refers to an abelian group which is the direct product of a cyclic group of order  $p^{m}$  and a cyclic group of order  $p^{n}$

### CHAPTER I

#### FUNDAMENTAL CONCEPTS

In this chapter the relationships between the classes of groups under investigation are established, and essential background material is provided. It should be emphasized that minimal nonabelian groups are called Redei groups and that only finite p-groups for primes p > 2 are considered.

Definition 1.1: Class 3 is the collection of all nonabelian groups G which satisfy the property:

(1.1)  $Z(R) = C_G(R)$  for each Redei subgroup R of G .

Definition 1.2: Class R is the collection of all nonabelian groups G which satisfy the property:

(1.2) R =  $C_G(Z(R))$  for each Redei subgroup R of G .

Proposition 1.1:  $\mathcal{R} \subseteq \mathcal{J}$  . Moreover, both  $\mathcal{R}$  and  $\mathcal{J}$  are subgroup inherited.

Proof:  $C_G(\mathbb{N}) \leq C_G(\mathbb{N})$  when  $\mathbb{N} \leq \mathbb{M} \leq \mathbb{G}$ . In particular  $C_G(\mathbb{R}) \leq C_G(\mathbb{Z}(\mathbb{R}))$  for each Redei subgroup  $\mathbb{R}$  of  $\mathbb{G}$ . If  $\mathbb{G} \in \mathcal{R}$ , then  $C_G(\mathbb{Z}(\mathbb{R})) = \mathbb{R}$ . Thus  $C_G(\mathbb{R}) \leq \mathbb{R}$ . Hence  $\mathbb{G} \in \mathcal{J}$ .

Let N be a nonabelian subgroup of G  $\epsilon$  R . Let R be a Redei subgroup of N . R is also a Redei subgroup of G , so  $C_{\rm G}({\rm Z(R)}) = {\rm R} \; . \; \; {\rm Thus} \; \; C_{\rm N}({\rm Z(R)}) = {\rm N} \cap C_{\rm G}({\rm Z(R)}) = {\rm N} \cap {\rm R} = {\rm R} \; . \; \; {\rm Hence} \; {\rm N} \; \epsilon \; {\cal R} \; . \; \; {\rm This} \; {\rm proves} \; {\rm that} \; \; {\cal R} \; \; {\rm is} \; {\rm subgroup} \; {\rm inherited}.$ 

The proof that 3 is subgroup inherited is similar.

Some basic properties of Redei subgroups are established in the next two results.

Proposition 1.2: If R is a Redei group, then

- (a) Z(R) has index  $p^2$  in R, and
- (b)  $Z(R) = \Phi(R)$ .

Proof: Each maximal subgroup of R is abelian. If there is a maximal subgroup A of R such that  $Z(R) \not\leq A$ , then G = AZ(R). Consequently, G is abelian; a contradiction is reached. Thus  $Z(R) \leq \Phi(G)$ .

Let M and M\* be any two maximal subgroups of R.  $R = MM^*$ . This implies that  $M \cap M^*$  has index  $p^2$  in R and that  $M \cap M^* \leq Z(R)$ . Hence  $Z(R) = \Phi(R) = M \cap M^*$ .

THEOREM 1.1: If G is a nonabelian group which is not Redei, then G has at least two Redei subgroups.

Proof: Let R be a Redei subgroup of G . Let  $M \leq G$  such that (M:R) = p. If M has at least two Redei subgroups, then G has at least two Redei subgroups. Therefore, without loss of generality, it can be assumed that R is a maximal subgroup of the group G .

Suppose, further, that R is the only Redei subgroup of G. Since G is not a cyclic group, it has at least two maximal subgroups. Also, the number of subgroups of a given order is congruent to 1. modulo p [15, p. 314]. From the fact that p > 2, it follows that the number of maximal subgroups of G is at least three. In particular, G has at least two abelian maximal subgroups A and  $A^*$ .  $G = AA^*$  so that  $A \cap A^* \leq Z(G)$ . However,  $(G : Z(G)) \geq p^2$  which implies that  $Z(G) = A \cap A^*$ . Therefore, G/Z(G) has order  $p^2$ . Now G/Z(G) has at least two abelian subgroups A/Z(G) and  $A^*/Z(G)$  of order p. Hence, G/Z(G) is elementary abelian, that is, abelian of type (1,1). This implies that G/Z(G) has p+l subgroups of order p. Thus, G has at least p+l abelian maximal subgroups, each corresponding to a subgroup of G/Z(G). But each abelian maximal subgroup contains Z(G)(otherwise G itself would be abelian), so G has exactly p+l abelian maximal subgroups. Thus, G has p+2 maximal subgroups in all, which contradicts the fact that the number must be congruent to 1, mod p. Hence, G must have at least two Redei subgroups.

For each group in class  ${\cal R}$  , there is a special relationship between its Redei subgroups and its maximal abelian subgroups.

# THEOREM 1.2: If $G \in \mathcal{R}$ , then

- (a) each maximal abelian subgroup of G is contained in a Redei subgroup of G, and
- (b) each maximal subgroup of a Redei group is a maximal abelian subgroup of G.

Proof: (a) Let A be a maximal abelian subgroup of G. Let  $M \leq G$  such that (M:A) = p. Since A is a maximal abelian subgroup of G, M is nonabelian. M contains a Redei subgroup R. M = AR so that  $p = (M:A) = (R:A \cap R)$ . By Proposition 1.1,  $Z(R) \leq A \cap R \leq A$ . This implies that  $A \leq C_G(Z(R)) = R$ . Thus M = R. Therefore, each maximal abelian subgroup of G is contained in a Redei subgroup.

(b) Let R be a Redei subgroup of G and let M be a maximal subgroup of R . Let A be a maximal abelian subgroup of G which contains M . By Proposition 1.1,  $Z(R) < M \le A$  . Thus  $A \le C_G(Z(R)) = R$  . So, M = A . Hence, each maximal subgroup of a Redei subgroup of G is a maximal abelian subgroup of G .

Theorem 1.2 can be improved to the following result.

THEOREM 1.3: If G  $\epsilon$  R , then there exists an integer  $i \geq 0$  such that each subgroup of G of index  $p^i$  is a Redei group.

Proof: The theorem is trivially satisfied for Redei groups, in which case i=0. Therefore, let  $G \in \mathcal{R}$  be a non-Redei group. By Theorem 1.2(a), G has no abelian maximal subgroup. Inductively, assume that each proper nonabelian subgroup of G satisfies the theorem. This is possible since  $\mathcal{R}$  is subgroup inherited.

Let R be a Redei subgroup of G. Let N be a subgroup of G such that (G:N) = (G:R). If there is a maximal subgroup M of G which contains both N and R, then by the inductive hypothesis N is a Redei subgroup. Assume, therefore, that there is no maximal subgroup

containing both R and N. Let M and M\* be maximal subgroups of G such that  $R \leq M$ ,  $N \leq M^*$ ,  $R \not\subseteq M^* \cap M$  and  $N \not\subseteq M^* \cap M$ . Let  $N^* \leq M^* \cap M$  such that  $(G:N^*) = (G:R)$ . The inductive hypothesis applied to M yields that  $N^*$  is a Redei subgroup of G. Then, from the application of the inductive hypothesis to  $M^*$ , it follows that N is a Redei group. This completes the proof.

In the above theorem, once the index of one Redei subgroup is known for a group in class R, it follows that each subgroup with that same index is also Redei. Thus, property (1.2), by which class R is defined, leads to another condition on Redei subgroups.

Definition 1.3: Class  $3^*$  is the collection of all nonabelian groups G which satisfy the property:

(1.3) If R is a Redei subgroup of G and  $(G:R) = p^{i}$ , then each subgroup of index  $p^{i}$  is a Redei subgroup of G.

Proposition 1.3:  $R \subseteq \mathcal{J}^* \subseteq \mathcal{J}$  . In addition, class  $\mathcal{J}^*$  is subgroup inherited.

Proof: Theorem 1.3 shows that  $R \subseteq 3$  \*.

Let G be a group in  $\mathcal{J}^*$ . Let R be a Redei subgroup of G and let M be a maximal subgroup of R . By property (1.3) M is a maximal abelian subgroup of G , so that  $M=C_G(M)$ . However, M < R implies that  $C_G(R) \leq C_G(M) = M < R$ . Hence G  $\epsilon$   $\mathcal{J}$ .

 $3^*$  is clearly subgroup inherited.

Proposition 1.4: Let G be a nonabelian group.

- (a) [4] If N is a normal subgroup of G of order  $p^2$ , then  $C_{G}(N)$  is a normal subgroup of index at most p.
- (b) If N is a normal abelian subgroup of type (2,1), then  $\mathcal{C}_{\mathbf{G}}(\mathbf{N})$  has index at most p in G .
- (c) If  $|G| = p^4$ , then G has an abelian maximal subgroup.

Proof: (a) Each element x of G generates an automorphism  $\tau_{\rm X}$  of N since N is normal in G . The mapping  ${\rm X} + \tau_{\rm X}$  is a homomorphism of G into Aut (N) , the group of automorphisms of N , with kernel  $C_{\rm G}({\rm N})$  . Thus,  $C_{\rm G}({\rm N})$  is normal in G and  ${\rm G}/{\rm C}_{\rm G}({\rm N})$  is isomorphic to a subgroup A of Aut (N) . Therefore, A is a p-group; the order of A divides the order of Aut (N) . Now |Aut (N)| = p(p-1) if N is elementary abelian. Hence |A| divides p and  ${\rm G}: C_{\rm G}({\rm N}) \leq {\rm p}$ .

- (b) By a result of Miller [18], Aut (2,1) is isomorphic to Aut (1,1) . As in (a),  $G/C_G(N)$  is isomorphic to a subgroup A of Aut (N) . Thus |A| divides p and  $(G\colon C_G(N))\leq p$  .
- maximal subgroup. Suppose, therefore, that G is not a Redei group. If Z(G) has order  $p^2$ , then there is a maximal subgroup M containing Z(G) and (M: Z(G)) = p . It follows that M is abelian. On the other hand, if |Z(G)| = p, then there is a normal subgroup N of G of order  $p^2$  with Z(G) < N . From part (a), (G:  $C_G(N)$ ) = p . But then,  $(C_G(N): N) = p$ . However,  $N \leq Z(C_G(N))$  which implies that  $C_G(N)$  is abelian. Hence G always has an abelian maximal subgroup.

Both inclusions in part (a) of Proposition 1.3 are, in fact, proper inclusions. Part (c) of Proposition 1.4 indicates that a group in  $\mathcal{J}^* \sim \mathcal{R}$  has order greater than  $p^4$ . An example of such a group is given in Chapter IV. The following example shows that  $\mathcal{J}^* \subset \mathcal{J}$ .

Example 1.1: If  $G = C_3 \ 1 \ C_3$ , the standard restricted wreath product of the cyclic group of order 3 by the cyclic group of order 3, then [2, p. 30]  $|G| = 3^{4}$  and  $|Z(G)| = |C_3| = 3$ . By Proposition 1.1, G is not Redei. By Proposition 1.4(c), G has an abelian maximal subgroup. Thus  $G \not\in \mathcal{J}^*$ .

Suppose that R is a Redei subgroup of G for which  $C_G(R) \not R$ .  $|R| = 3^3$  and  $G = RC_G(R)$ . Since R is nonabelian,  $C_G(R) < G$ . Now  $R \cap C_G(R) \le Z(G)$ . Also,  $3 = (G:R) = (C_G(R):R \cap C_G(R))$ . It then follows that  $|C_G(R)| = 3^2$ , so that  $C_G(R)$  is abelian. Hence,  $C_G(R) \le Z(G)$ ; a contradiction is reached. Therefore, for each Redei subgroup R,  $C_G(R) < R$ . Consequently,  $G \in \mathcal{J}$ .

The rest of the chapter contains necessary background material.

If the result is known, its proof has been omitted.

THEOREM 1.4: [20] If G is a group for which each proper subgroup is abelian, then  $G = \langle a, b \rangle$  has defining relations:

(1.4) 
$$a^{p^m} = b^{p^n} = 1$$
,  $b^{-1}ab = a^{1+p^{m-1}}$   $(m \ge 2, n \ge 1)$ ; or

(1.5) 
$$a^{p^m} = b^{p^n} = c^p = 1$$
,  $b^{-1}ab = ac$ .

If G satisfies (1.4), then  $|G|=p^{m+n}$ . If G satisfies (1.5), then  $|G|=p^{m+n+1}$ .

Proposition 1.5: If [a,b] commutes with a, then  $[a^n,b]$  =  $[a,b]^n$  for every integer n.

## Definition 1.4: A group is called regular if

 $(xy)^p = x^p y^p I d_1^p$  for each pair x,  $y \in G$  where  $d_1 \in \langle x, y \rangle'$ .

THEOREM 1.5: If G is a Redei group, then

- (a) G has order p,
- (b) G is a regular group.

Proof: (a) Let x,  $y \in G$ .  $\phi(G) = G' \upsilon_1(G)$  [15, p. 272]. By Proposition 1.1,  $G' \leq Z(G)$ . By Proposition 1.5,  $[x,y]^p = [x^p,y]$ . However,  $\upsilon_1(G) \leq Z(G)$  so that  $[x^p,y] = 1$ . Hence |G'| = p.

(b) Because G' is cyclic, it can be concluded that G is regular [15, p. 322].

THEOREM 1.6: [13] If G is a regular group, then

- (a)  $|G/\Omega_{\uparrow}(G)| = |\upsilon_{\uparrow}(G)|$ ,
- (b) each element of  $\,\upsilon_1(G)\,$  is the  $p^{\,{\rm th}}$  power of an element of  $\,G\,$  and each element of  $\,\Omega_1(G)\,$  has order  $\,p\,$  .

THEOREM 1.7: [13] (a) If the class of G is less than p, then G is a regular group.

(b) If G is a two-generator 3-group in which G' is not cyclic, then G is not a regular group.

THEOREM 1.8: Let G be a Redei group; then

- (a)  $v_1(G) = \Phi(G)$  when G has defining relations (1.4),
- (b)  $(\Phi(G): \upsilon_1(G)) = p$  and  $G' \cap \upsilon_1(G) = 1$  when G has defining relations (1.5).

<u>Definition 1.5</u>: A group G is called metacyclic if and only if there exists a normal subgroup N such that both N and G/N are cyclic.

THEOREM 1.9: [16] G is a metacyclic group if and only if  $|G/\upsilon_1(G)| \leq p^2 \ .$  Moreover, each metacyclic group is a regular group.

By reason of Proposition 1.2 and Theorem 1.8, a Redei group with defining relations (1.4) satisfies Theorem 1.9, so it is a metacyclic group.

THEOREM 1.10: [3] Let G be a group for which all proper subgroups are metacyclic but G itself is not. Then, G is one of the following.

- (a) G is elementary abelian of order  $p^3$  .
- (b) G is the nonabelian group of order  $p^3$  and exponent p .
- (c) G is a 3-group of class 3 and order 3<sup>4</sup>.

Proposition 1.6: [5] If G is a nonabelian group with two generators, then  $\Phi(G')G_3$  is the only maximal subgroup of G' which is normal in G .

Proposition 1.7: [5] If G is a nonmetacyclic group with two generators, then  $G/\Phi(G^{\dagger})G_3$  has defining relations [a,b]=c,  $a^{pm}=b^{pn}=c^p=1$ , [a,c]=[b,c]=1 in terms of two generators a,b where  $G/G^{\dagger}$  is abelian of type (m,n). This factor group is the Redei group with defining relations (1.5).

### CHAPTER II

### REDEI SUBGROUPS AS CENTRALIZERS OF THEIR CENTERS

The relation  $C_G(N) \leq C_G(R)$  for  $R \leq N \leq G$  implies that property (1.1) which defines class  $\mathcal J$  can be extended to all nonabelian subgroups of G; that is,

Z(N) =  $\mathcal{C}_G$ (N) for each nonabelian subgroup N of G if and only if  $Z(R) = \mathcal{C}_G(R) \text{ for each Redei subgroup R of G .}$ 

The question is raised as to whether or not property (1.2) which defines class R can also be extended to all nonabelian groups. An affirmative answer is given in this chapter. Moreover, the structure of the groups in class R is completely determined.

Definition 2.1: Class  $\mathcal{R}^{*}$  is the collection of all nonabelian groups G which satisfy the property:

(2.1) N =  $C_G(Z(N))$  for each nonabelian subgroup N of G .

Property (2.1) is a stronger version of property (1.2). Thus  $R * \subseteq R$ . Also, each subgroup of a group in R \* is a group in R \*.

Proposition 2.1: If  $G \in R^*$ , then

- (a) Z(N) < Z(M) if  $M < N \le G$  and M is nonabelian,
- (b)  $Z(M) \neq Z(N)$  if (G:M) = (G:N) and  $M \neq N$ .

Proof: (1) Since Z(M) < M < N , then Z(N)  $\leq$   $C_G(Z(M)) = M$  . Thus Z(N)  $\leq$  Z(M) . If Z(N) = Z(M) , then M =  $C_G(Z(M)) = C_G(Z(N)) = M$ ; a contradiction is reached. Hence Z(N) < Z(M) .

(2)  $\mathcal{R}^*\subseteq\mathcal{R}$ . By Theorem 1.3, M and N are both abelian or both nonabelian. If both are abelian, then  $Z(M)=M\neq N=Z(N)$ . On the other hand, suppose that both are not abelian. If Z(M)=Z(N), then  $M=C_G(Z(M))=C_G(Z(N))=N$ , which contradicts  $M\neq N$ . Thus,  $Z(M)\neq Z(N)$  when neither M nor N is abelian.

Closer examination of part (a) of Proposition 2.1 yields the following stronger result.

THEOREM 2.1: G  $\epsilon$  R \* if and only if (Z(M): Z(N)) = (N:M) whenever M is nonabelian and M  $\leq$  N  $\leq$  G .

Proof: Assume that  $(Z(M)\colon Z(N))=(N:M)$  whenever M is a nonabelian subgroup of G and  $M\le N\le G$ . Let B< G.  $B\le C_G(Z(B))$ . By hypothesis,  $Z(C_G(Z(B)))\le Z(B)$ . However,  $Z(B)\le B\le C_G(Z(B))$  so that  $Z(B)\le Z(C_G(Z(B)))$ . Thus,  $Z(B)=Z(C_G(Z(B)))$ . According to the hypothesis, it follows that  $B=C_G(Z(B))$ . Hence,  $G\in \mathcal{R}^*$ .

The proof for the other direction will proceed by induction on the index of the Redei subgroups in G . This is possible by Theorem 1.3.

Let  $G \in R^*$ . If G is a Redei group, then the condition is trivially satisfied. Therefore, let G be a non-Redei group and let  $(G:R) = p^1$ , i > 0, for each Redei subgroup R of G. By induction, assume that any group in  $R^*$ , for which the index of the Redei subgroups is less than  $p^1$ , satisfies the theorem in the direction under consideration.

Let M be a nonabelian subgroup of G and let  $M \le N \le G$ . If N < G, then  $(N:R) < p^1$  where R is a Redei subgroup of N . N  $\in \mathcal{R}^*$ ; so, by the induction hypothesis,  $(Z(M)\colon Z(N))=(N:M)$ . It remains to show that  $(Z(M)\colon Z(G))=(G:M)$ . Let B be a maximal subgroup of G such that  $M \le B$ . B  $\in \mathcal{R}^*$ .  $(B:R) < p^1$  for each Redei subgroup R of B. Then,  $(Z(M)\colon Z(B))=(B:M)$  by the inductive hypothesis. By Proposition 2.1(a), Z(G) < Z(B). It must next be shown that  $(Z(B)\colon Z(G))=p$ ; then,

$$(Z(M): Z(G)) = (Z(M): Z(B))(Z(B): Z(G)) = (B:M)(G:B) = (G:M)$$

which will complete the proof.

Let B be a maximal subgroup of G . Since G is not cyclic, G has another maximal subgroup  $B^*$ . By Theorem 1.3, both B and  $B^*$  are nonabelian. By Proposition 2.1,  $Z(B) \neq Z(B^*)$ , Z(G) < Z(B) and  $Z(G) < Z(B^*)$ . Thus,

$$Z(G) \leq Z(B) \cap Z(B^*).$$

Two cases arise. In the first case, B and B\* are themselves Redei groups. Then, by Proposition 1.1,  $Z(B) = \Phi(B)$  and  $Z(B^*) = \Phi(B^*)$ , each having index  $p^2$  in B and B\*, respectively. Thus, (G: Z(B)) =

 $(G: Z(B^*)) = p^3$ . Now  $\phi(B) \leq \phi(G) \leq B \cap B^*$ , and  $\phi(B^*) \leq \phi(G) \leq B \cap B^*$  [15, p. 273]. Since  $Z(B) \neq Z(B^*)$ , it follows that  $\phi(B) < \phi(G)$  and that  $\phi(B^*) < \phi(G)$ . These, in turn, imply that  $\phi(G) = B \cap B^*$ . Then,  $\phi(G)$  has index  $p^2$  in G; so,  $\phi(G)$  is a maximal abelian subgroup of G. Also,

$$Z(B^*) < Z(B)Z(B^*) = \Phi(B)\Phi(B^*) < \Phi(G)$$
.

Consequently,  $Z(B)Z(B^*) = \Phi(G)$ . Thus,

$$p = (\Phi(G): Z(B^*)) = (Z(B): Z(B) \cap Z(B^*))$$
.

However,  $G = BB^*$ . Thus,  $Z(B) \cap Z(B^*) \le Z(G)$ . By (2.2),  $Z(G) = Z(B) \cap Z(B^*)$ . Hence, (Z(B): Z(G)) = p.

In the second case, B and B\* are not Redei groups. Then,  $B \cap B^*$ , which has index  $p^2$  in G, is not abelian. By the inductive hypothesis,  $(Z(B \cap B^*): Z(B^*)) = (B: B \cap B^*) = p$ . Since  $Z(B) \neq Z(B^*)$ , then  $Z(B \cap B^*) = Z(B)Z(B^*)$ . Therefore,  $(Z(B): Z(B) \cap Z(B^*)) = p$ . But  $G = BB^*$  which implies that  $Z(B) \cap Z(B^*) \leq Z(G)$ . By (2.2)  $Z(G) = Z(B) \cap Z(B^*)$ . Hence (Z(B): Z(G)) = p.

The proof is now complete.

Each Redei group is a member of both  $\mathcal{R}^*$  and  $\mathcal{R}$ . If  $G \in \mathcal{R}$  and if G has a maximal subgroup that is a Redei subgroup, then by Theorem 1.3, each maximal subgroup is a Redei subgroup. Consequently,  $G \in \mathcal{R}^*$ . Thus, the restrictions of class  $\mathcal{R}$  and class  $\mathcal{R}^*$  to their minimal non-Redei groups coincide. The next theorem describes these groups and provides the initial step toward the classification of groups in  $\mathcal{R}$ .

THEOREM 2.2: If G  $\epsilon$  R and if G is minimal with the property that it is not a Redei group, then G is metacyclic with order greater than  $p^5$  .

Proof: Let R be a Redei subgroup of G . If (G:R) > p, then there exists a maximal subgroup M of G containing R . However, M  $\in$  R and M is not Redei, which contradicts the fact that G is a minimal non-Redei group. Thus, each Redei subgroup of G is a maximal subgroup. By the remarks preceding the theorem each maximal subgroup of G is a Redei subgroup, and G  $\in$  R  $^*$ . By Proposition 1.2,  $(G: Z(R)) = p^3$  and  $Z(R) = \Phi(R)$  for each Redei subgroup R of G . By Theorem 2.1,

(2.3) 
$$(G: Z(G)) = p^{4}$$

Also, from Proposition 2.1, distinct Redei subgroups have distinct Frattini subgroups. It follows that

(2.4) 
$$(G: \Phi(G)) = p^2,$$

so that  $\Phi(G)$  is a maximal abelian subgroup of  $\,G\,$  .

Proposition 1.4(c) implies that  $|G| > p^4$ . If  $|G| = p^5$ , then, from (2.3) and (2.4), |Z(G)| = p,  $Z(R) = p^2$  for each Redei subgroup R of G, and  $|\Phi(G)| = p^3$ . Since  $\Phi(G)$  is maximal abelian in G,  $\Phi(G) = C_G(\Phi(G))$ .  $\Phi(G)$  contains the distinct centers of the maximal subgroups so that  $\Phi(G)$  is not cyclic. If  $\Phi(G)$  is abelian of type (2,1), then by Proposition 1.4(b),  $(G:C_G(\Phi(G))) = p$ . Thus,  $\Phi(G) < C_G(\Phi(G))$  which is a contradiction to the fact that  $\Phi(G)$  is a maximal abelian subgroup of G. Hence,  $\Phi(G)$  is elementary abelian; so,  $\Phi(G) \leq \Omega_1(G)$ .

Now  $\Phi(G)$  is a maximal subgroup of each Redei subgroup R of G . Thus  $|\Omega_1(R)| \geq p^3$  . From the fact that R is regular (Theorem 1.5) and from Theorem 1.6, it follows that  $|\upsilon_1(R)| \leq p$  . However, as a result of Theorem 1.8  $|\upsilon_1(R)| \geq p$  . Hence,  $|\upsilon_1(R)| = p$  . Moreover, by Theorem 1.5, |R'| = p . Both  $\upsilon_1(R)$  and R', as characteristic subgroups of R, are normal in G. This implies that both  $\upsilon_1(R)$  and R' intersect Z(G) nontrivially. Therefore, Z(G) =  $\upsilon_1(R) = R'$ . As a consequence of  $\Phi(G) = G'\upsilon_1(G)$ ,  $|\Phi(G)| = p$ ; this is a contradiction to  $|\Phi(G)| = p^3$ . Hence,  $|G| > p^5$ .

The proof that G is metacyclic will be by contradiction. Therefore, suppose that G is not metacyclic. By Theorem 1.10 there exists a subgroup of G that is not metacyclic. Because the metacyclic property is subgroup inherited, there is then a maximal subgroup R of G that is not metacyclic. By Theorem 1.9,  $(R: \upsilon_1(R)) > p^2$ . Since R is Redei, then from Proposition 1.2,  $(R: \varphi(R)) = p^2$ . From Theorem 1.9, |R'| = p. These last two relations imply that  $|R'| \cap |U_1(R)| = 1$ , so  $|R: \upsilon_1(R)| = p^3$ . Moreover, |R'| is characteristic in R which implies that  $|R'| \leq |Z(G)|$ . If  $|U_1(R)| \leq |Z(G)|$ , then  $|Z(R)| = |\varphi(R)| \leq |Z(G)|$ ; this is a contradiction to Proposition 2.1. Hence,

(2.5) 
$$v_1(R) \not\leq Z(G)$$
.

Suppose that  $\upsilon_1(G) = \upsilon_1(R)$ . From Theorem 1.8, it follows that  $(M: \upsilon_1(M)) \leq p^3$  for each Redei subgroup M of G. Thus  $\upsilon_1(G) = \upsilon_1(M)$  for each maximal subgroup M of G. This implies that  $\upsilon_1(G) \leq Z(M)$  for each maximal subgroup M of G. By Theorem 2.1, (Z(M): Z(G)) = p for each maximal subgroup of G. Then, Z(G) is the intersection of the centers of the maximal subgroups; so  $\upsilon_1(G) \leq Z(G)$ . However, this

contradicts (2.5). So,

(2.6) 
$$v_1(R) < v_1(G)$$
.

Now from Theorem 1.9 (G:  $\upsilon_1(G)$ ) >  $p^2$ . However, (G:  $\upsilon_1(R)$ ) =  $p^4$ . Consequently,

(2.7) (G: 
$$v_1(G)$$
) =  $p^3$ .

Suppose that  $Z(G) < \upsilon_1(G)$ . M' is a characteristic subgroup of M for each maximal subgroup M of G which implies that  $M' \leq Z(G)$  for each maximal subgroup M of G. Then,  $Z(M) = \Phi(M) = M'\upsilon_1(M) \leq \upsilon_1(G)$  for each maximal subgroup M. But  $(M: Z(M)) = p^3$  so that  $Z(M) = \upsilon_1(G)$ . This is a contradiction to Proposition 2.1(b). Hence,  $Z(G) \not\prec \upsilon_1(G)$ . Consequently,  $\upsilon_1(G)$  is not the center of any maximal subgroup. Furthermore,

(2.8) 
$$M' \not\sim \upsilon_1(G)$$
 for each maximal subgroup M of G.

Assume that there are two maximal subgroups  $R_1$  and  $R_2$  such that  $R_1' \neq R_2'$ . Then  $|R_1'R_2'| = p^2$ . Denote  $R_1'R_2'$  by A. If G/A is abelian, then G' < A. In particular, M' < A for each maximal subgroup M of G. Suppose that G/A is not abelian. G/A has two abelian maximal subgroups  $R_1/A$  and  $R_2/A$ . Thus,  $Z(G/A) = R_1/A \cap R_2/A$  and Z(G/A) has index  $p^2$  in G/A. Also,  $\Phi(G/A) \leq Z(G/A)$ . However,  $G/A: \Phi(G/A) = p^2$  since

$$(G/A: \Phi(G/A)) = (G/A: \Phi(G)/A) = (G: \Phi(G)) .$$

Hence,  $Z(G/A) = \Phi(G/A)$ . It then follows that each maximal subgroup of G/A is abelian. Therefore  $M' \leq A$  for each maximal subgroup M of G.

Suppose that the commutators are distinct for distinct maximal subgroups. Since  $(G: \phi(G)) = p^2$ , then G has p+1 maximal subgroups. However, A is elementary abelian of order  $p^2$ , so that A has p+1 subgroups of order p. Thus each subgroup of A of order p is the commutator subgroup of a maximal subgroup. From (2.8) it follows that  $A \cap \upsilon_1(G) = 1$ . However, (2.4) and (2.7) imply that  $A \cap \upsilon_1(G) \neq 1$ ; a contradiction is reached. Thus G has at least two maximal subgroups  $M_1$  and  $M_2$  for which  $M_1' = M_2'$ . Denote this subgroup  $M_1'$  by  $A^*$ . By assumption,  $R_1' \neq R_2'$ . Thus  $G/A^*$  is not abelian. However,  $G/A^*$  has two abelian maximal subgroups  $M_1/A^*$  and  $M_2/A^*$ . Therefore  $Z(G/A^*)$  has index  $p^2$  in  $G/A^*$  and  $Z(G/A^*) = \phi(G/A^*)$ . Consequently each maximal subgroup of  $G/A^*$  is abelian. In particular,  $R_1' = R_2' = A$ ; a contradiction is reached. Hence, the commutator subgroups of the maximal subgroups are equal. Denote this subgroup of G of order P by  $R^*$ .

If  $G'=R^*$ , then  $G'\leq Z(G)$ . By Proposition 1.5,  $[x^p,y]=[x,y]^p$ . But  $[x,y]^p=1$ . It follows that  $\upsilon_1(G)\leq Z(G)$ . Then  $\varphi(G)\leq Z(G)$  which contradicts (2.3) and (2.4). Thus  $|G'|\geq p^2$ . Now each maximal subgroup of  $G/R^*$  is abelian, but  $G/R^*$  is not abelian; so,  $G/R^*$  is a Redei group. Hence,  $|G'|=p^2$ . But then, from (2.2) and (2.7),  $|G'\cap \upsilon_1(G)|=p$ .

Suppose  $G' \not\prec Z(G)$ .  $R^* \leq Z(G)$ , which implies that  $G' \cap Z(G) = R^*$ . In addition,  $G' \cap \upsilon_1(G) \triangleleft G$ . Thus  $G' \cap \upsilon_1(G) \leq Z(G)$ . Therefore,  $G' \cap \upsilon_1(G) = G' \cap Z(G) = R^*$ . Thus  $M' < \upsilon_1(G)$  for each maximal subgroup M, which contradicts (2.8). Thus G' < Z(G).

By Proposition 1.5,  $[x,y]^p = [x^p,y]$ . If G' is elementary abelian, then  $[x^p,y] = 1$ . Therefore,  $\upsilon_1(G) \leq Z(G)$ , which contradicts (2.6). G' is thus a cyclic subgroup. Then,  $R^* = G' \cap \upsilon_1(G)$  which contradicts (2.8).

Since all possible avenues have been exhausted, it follows that the group G must be a metacyclic group. This completes the proof.

Example 2.1: Given  $G = \langle a,b | a^{p^4} = b^{p^2} = 1$ ,  $[a,b] = a^{p^2} \rangle$ . This group is used to illustrate Theorem 2.2 and to show that there are groups in class R which are not Redei groups.

For this group,  $|G|=p^6$ ;  $\upsilon_1(G)=\langle a^p\rangle\langle b^p\rangle$ , and  $\upsilon_1(G)$  has index  $p^2$  in G. By Theorem 1.9, G is metacyclic. Also,  $G'=\langle ap^2\rangle$  and  $G'\leq Z(G)$ .

Let  $M=\langle a > \langle b^p \rangle$ . Let  $M^*=\langle a^p > \langle b \rangle$ . The relations  $b^{-p}ab^p=a^{1+p^3}$  and  $b^{-1}a^pb=a^{p+p^3}$  imply that  $M'=M^{*'}=\langle a^p^3 \rangle$ . Let  $A=\langle a^p^3 \rangle$ . G/A is metacyclic but is not abelian. M/A and  $M^*/A$  are abelian maximal subgroups, so that  $Z(G/A)=M/A\cap M^*/A=\Phi(G/A)$ . Each maximal subgroup of G/A is abelian; A is the commutator subgroup of each nonabelian maximal subgroup. Because Z(G) is a characteristic subgroup,  $Z(G) \leq \Phi(G)$ ; in particular,  $Z(G) \leq Z(M)$  and  $Z(G) \leq Z(M^*)$ . But  $Z(M)=\langle a^p \rangle$  and  $Z(M^*)=\langle a^p \rangle$  so that  $Z(G) \leq \langle a^p \rangle$ . Hence  $Z(G)=\langle a^p \rangle = G'$ .

Suppose that G has an abelian maximal subgroup  $A^*$ . Since  $Z(M) \leq \Phi(G)$ , then  $Z(M) < A^*$ . Thus,  $Z(M) \leq Z(G)$ , which contradicts the fact that  $Z(G) = \langle ap^2 \rangle \langle Z(M) \rangle$ . Therefore, each maximal subgroup of G is nonabelian.

Since  $B' = A \le Z(G)$  for each maximal subgroup B , then in view

of Proposition 1.5  $[x^p,y] = [x,y]^p = 1$ . It follows that  $\upsilon_1(B) \leq Z(B)$ , and consequently,  $\Phi(B) \leq Z(B)$ . Since G is metacyclic, then B is metacyclic and  $(B:\Phi(B)) = p^2$ . Therefore,  $Z(B) = \Phi(M)$  which implies that each maximal subgroup of B is abelian. B is thus a Redei subgroup. Moreover, Z(G) < Z(B) which leads to  $B = C_G(Z(B))$ . Hence,  $G \in \mathcal{R}$ .

THEOREM 2.3: If G  $\epsilon$  R and G is not a Redei group, then G is metacyclic.

Proof: By Theorem 1.3, an inductive proof on the index of the Redei subgroups can be applied. Let R be a Redei subgroup of G and let  $(G:R)=p^k$ . Since G is not a Redei group,  $k\ge 1$ . If k=1, then G is metacyclic by Theorem 2.2. Also,  $|G|\ge p^6$ . Therefore, let k>1. Let M be a maximal subgroup of G. M  $\epsilon$  R. For each Redei subgroup R of M,  $(M:R)=p^{k-1}$ . By the inductive hypothesis, M is a metacyclic group. Hence, each proper subgroup of G is metacyclic. Because of Theorem 2.2,  $|G|\ge p^7$ . Therefore, it follows from Theorem 1.10 that G is itself metacyclic.

The above theorem places all non-Redei groups of class  ${\cal R}$  in the class of metacyclic groups. The next theorem leads to the reverse inclusion.

THEOREM 2.4: If G is a nonabelian metacyclic p-group, then G  $\epsilon$   $\ensuremath{\mbox{\it R}}^{\mbox{\tt \#}}$  .

Proof: Let G be a counterexample of minimal order. G is not a Redei group and  $|G| \ge p^{\frac{14}{9}}$ . There exists a subgroup  $M_0$  such that  $M_0 < \mathcal{C}_G(Z(M_0))$ .

Suppose that G has a nonabelian subgroup  $N_0$  such that  $(G:N_0) > p$ . For any nonabelian subgroup N of G such that (G:N) > p, there is a maximal subgroup N\* such that  $N < N^*$ . N\* is metacyclic. By hypothesis, N\*  $\epsilon$   $R^*$ . Thus,  $N = C_{N^*}(Z(N)) = N^* \cap C_{G}(Z(N))$ . If  $C_{G}(Z(N)) = G$ , then  $N = N^*$ ; a contradiction has been reached. Therefore,  $C_{G}(Z(N)) < G$ . Denote  $C_{G}(Z(N))$  by B. B is metacyclic and nonabelian. By hypothesis, B  $\epsilon$   $R^*$ ; so,  $N = C_{G}(Z(N)) = C_{G}(Z(N)) \cap B = B$ . Thus,

(2.9)  $N = C_{G}(Z(N))$  whenever N is nonabelian and (G:N) > p .

In particular,  $N_0 = C_G(Z(N_0))$  . Furthermore  $(G:M_0) = p$  . Thus,  $G = C_G(Z(M_0))$  which implies that

$$(2.10) Z(M_0) \le Z(G) .$$

By Theorem 1.9,  $(G: \Phi(G)) = p^2$ . If  $N_0^*$  is a maximal subgroup of G containing  $N_0$ , then  $N_0^* \in \mathcal{R}^*$  and  $(N_0^*: \Phi(G)) = p < (N_0^*: N_0)$ . By Theorem 1.3,  $\Phi(G)$  is nonabelian. By (2.9),  $\Phi(G) = C_G(Z(\Phi(G)))$ . Thus  $Z(G) \leq \Phi(G) < M_0$ . From (2.10) it follows that  $Z(M_0) = Z(G)$ .

Let M be any maximal subgroup of G other than  $M_0$ .  $Z(M_0) = Z(G) < \Phi(G) < M \text{ which implies that } Z(M_0) \leq Z(M) \text{ . Now M } \epsilon \text{ $\ell$}^*$  since  $\Phi(G)$  is nonabelian. But then, by Theorem 2.1,  $(Z(\Phi(G))\colon Z(M)) = p \text{ . In the same way, } (Z(\Phi(G))\colon Z(M_0)) = p \text{ . Thus}$ 

 $Z(M) = Z(M_0)$  . Therefore,

(2.11) Z(G) = Z(M) for each maximal subgroup M of G.

From Theorems 1.9 and 1.6 it follows that  $|\Omega_1(G)| = p^2$ . If  $\Omega_1(G) \not \in Z(G)$ , then  $C_G(\Omega_1(G))$  is a maximal subgroup of G by Proposition 1.4(a). By (2.11)  $Z(C_G(\Omega_1(G))) = Z(G)$ . However,  $\Omega_1(G) < Z(C_G(\Omega_1(G)))$ . Thus,  $\Omega_1(G) < Z(G)$  which is a contradiction. Therefore,  $\Omega_1(G) \le Z(G)$ , and accordingly, Z(G) is not cyclic.

Now G' is cyclic since G is metacyclic. There is then a subgroup A of Z(G) such that |A|=p and  $A \not\sim G'$ . G/A is non-abelian and metacyclic. Since |G/A|<|G|,  $G/A \in \mathcal{R}^*$ . Let  $Z^* \leq G$  such that  $Z^*/A = Z(G/A)$ .  $[Z^*,G] \leq A \cap G' = 1$ , so that  $Z^* \leq Z(G)$ . Hence  $Z(G) = Z^*$ , that is, Z(G/A) = Z(G)/A.

Let M/A be a maximal subgroup of G/A . M/A is nonabelian. By Theorem 2.1,  $(Z(M/A)\colon Z(G/A))=p$ . Let  $M^{\bigstar}< M$  such that  $M^{\bigstar}/A=Z(M/A)$ . Thus  $(M^{\bigstar}\colon Z(G))=p$ . However,  $[M^{\bigstar},M]\le A\cap G'=1;$  so,  $M^{\bigstar}\le Z(M)$ . By (2.11),  $M^{\bigstar}\le Z(G);$  a contradiction is reached. Hence, G has no nonabelian subgroup of index greater than p. Consequently,  $M_0$  is a Redei group, and  $Z(M_0)\le Z(G)$ .

Since  $M_{\Omega}$  is nonabelian, then

(2.12) 
$$Z(G) < v_1(G) = \phi(G)$$
.

Thus,  $C_G(\upsilon_1(G)) < G$ .  $\upsilon_1(G)$  is abelian, however; and, this implies both that  $(C_G(\upsilon_1(G)) : \upsilon_1(G)) \le p$  and that  $C_G(\upsilon_1(G))$  is abelian. Let  $y \in C_G(\upsilon_1(G))$ . For each  $x \in G$ ,  $[x^p,y] = 1$ . Now G is regular so

that  $[x^p,y] = [x,y^p]$ , [12, p. 185]. Thus,  $v_1(c_G(v_1(G))) \le Z(G) .$ 

Suppose that  $\upsilon_1(C_G(\upsilon_1(G))) \leq Z(M_0)$ . Let  $g \in G$ .  $g \in M^*$  for some maximal subgroup  $M^*$  of G. If  $M^*$  is abelian, then  $M^* = C_G(\upsilon_1(G))$ . But then  $g^P \in \upsilon_1(M^*) = \upsilon_1(C_G(\upsilon_1(G))) \leq Z(M_0)$ . If, on the other hand,  $M^*$  is nonabelian, then  $g^P \in \upsilon_1(M^*) = \Phi(M^*) = Z(M^*)$ . Now  $Z(M_0) = \upsilon_1(M_0) < \Phi(G) < M^* < C_G(Z(M_0))$  so that  $Z(M_0) \leq Z(M^*)$ . Since both  $M_0$  and  $M^*$  are Redei groups,  $|Z(M_0)| = |Z(M^*)|$ . Thus  $Z(M_0) = Z(M^*)$  so that  $g^P \in Z(M_0)$ . Therefore,  $\upsilon_1(G) \leq Z(G)$ , which contradicts (2.12). Therefore,  $\upsilon_1(C_G(\upsilon_1(G))) \not\leq Z(M_0)$ . Since  $(\upsilon_1(G):\upsilon_1(M_0)) = p$ , it follows that  $\upsilon_1(G) = Z(M_0)\upsilon_1(C_G(\upsilon_1(G)))$ . Thus, from (2.13)  $\upsilon_1(G) \leq Z(G)$  which contradicts (2.12).

Since no other cases are possible, no such minimal counterexample exists.

A new characterization of nonabelian metacyclic p-groups is obtained as a corollary to the above theorem.

Corollary 2.4.1: If G is nonabelian and is not Redei, then G is metacyclic if and only if G  $\epsilon$  R and G has an abelian subgroup of order p  $^4$  .

Proof: If G is a nonabelian metacyclic group which is not a Redei group, then G  $\epsilon$  R  $^*\subseteq R$  by Theorem 2.4. Since G is not Redei, there exists a Redei subgroup R < G . Furthermore, there exists a subgroup R  $^*$  of G such that R < R  $^*$   $\leq$  G and (R  $^*$ : R) = p . R  $^*$  is a minimal non-Redei group which belongs to R . In view of Theorem 2.3,

 $|R^*| \ge p^6$  . Moreover, if M is a maximal subgroup of R , then M is abelian. But  $|M| \ge p^4$  . Hence G has an abelian subgroup of order  $p^4$  .

The converse is Theorem 2.3.

Corollary 2.4.2:  $R^* = R$ .

Proof: Let  $G \in \mathcal{R}$  . If G is either abelian or a Redei group, then  $G \in \mathcal{R}^*$ . Therefore, suppose that G is a non-Redei group in  $\mathcal{R}$ . By Theorem 2.3, G is a metacyclic group. By Theorem 2.4,  $G \in \mathcal{R}^*$ . Thus  $\mathcal{R} \subseteq \mathcal{R}^*$ .

If a group  $G \in \mathcal{R}$ , then Corollary 2.4.2 states that each non-abelian subgroup of G is a centralizer in G, namely, the centralizer of its center. This result supplements the investigation of Gaschütz [8] where he has dealt with groups, for which each subgroup is a centralizer.

Furthermore, Theorem 2.3 and Corollary 2.4.1 give a complete description of the groups in class R as either metacyclic or nonmetacyclic Redei. The metacyclic groups are described by defining relations in the following theorem.

THEOREM 2.5: [15] If G is a metacyclic group then G has the defining relations:

(2.14) 
$$a^{p^m} = 1$$
,  $b^{p^n} = a^{p^\ell}$ ,  $b^{-1}ab = a^k$ 

where  $\ell \ge 0$ ,  $k^{p^m} \equiv 1 \pmod{p^m}$  and  $p^{\ell}(k-1) \equiv 0 \pmod{p^m}$ . Furthermore,

$$|G| = p^{a+b}$$
.

A group in class  $\cal R$  has either the defining relations (1.5) or the defining relations (2.14).

## CHAPTER III

## REDEI SUBGROUPS WHICH CONTAIN THEIR CENTRALIZERS

Unlike the situation for a group from class R where all Redei subgroups have the same index, no relationship is known for the Redei subgroups of a group from class R. For this reason, attention is first focused upon groups from class R0 which have a Redei maximal subgroup. The particular groups that have an abelian maximal subgroup as well as a Redei maximal subgroup are characterized in this chapter.

Proposition 3.1: If  $G \in \mathcal{J}$  and G is not a Redei group, then G has at most one abelian maximal subgroup.

Proof: Suppose that G has two abelian maximal subgroups, A and A. From  $G = AA^*$ , it follows that  $Z(G) = A \cap A^*$  and that  $(G: Z(G)) = p^2$ . Since  $G \in \mathcal{J}$ , then  $Z(G) \leq Z(R)$  for each Redei subgroup R. However,  $(R: Z(R)) = p^2$ , for each Redei subgroup by Proposition 1.1. Thus, G = R, which is a contradiction. Hence, G has at most one abelian maximal subgroup.

Proposition 1.4(c) states that each nonabelian, non-Redei group of order  $p^4$  has at least one abelian maximal subgroup. The group in Example 1.1 is one such group, and it has been shown that this group is in fact an element in  $\beta$ . This group is, therefore, included in the following proposition.

Proposition 3.2: If  $|G| = p^4$  and if G has exactly one abelian maximal subgroup, then  $G \in \mathcal{J}$ .

Proof: Let A be the abelian maximal subgroup of G . Let M be any nonabelian maximal subgroup.  $|M|=p^3$  and M is a Redei group. Thus,  $M'=\Phi(M)=Z(M)$  with order p . Since A is abelian, Z(G)<A .

Suppose there exists an  $M_0$ , nonabelian, such that  $C_G(M_0) \neq Z(M_0)$ . Thus  $C_G(M_0) \neq M_0$ . If  $C_G(M_0)$  is a maximal subgroup of G, then  $G = M_0C_G(M_0)$  and  $M_0 \cap C_G(M_0)$  has order  $p^2$ . But  $M_0 \cap C_G(M_0)$  is then abelian, which forces  $M_0 \cap C_G(M_0) \leq Z(M_0)$ ; a contradiction is reached. Therefore,  $C_G(M_0)$  has index  $p^2$  in G. However,  $C_G(M_0)$  is then abelian. Since  $G = M_0C_G(M_0)$ , it follows that  $C_G(M_0) = Z(G)$ . Now, if N is any other subgroup of index  $p^2$  different from Z(G), then N is abelian. This forces Z(G)N to be abelian. Thus, A = Z(G)N and A contains every subgroup of index  $p^2$ . Furthermore, each nonabelian maximal subgroup contains exactly one maximal subgroup, its intersection with A. The nonabelian subgroup is then cyclic; and a contradiction is reached.

Therefore,  $C_{\rm G}({\rm M_0})={\rm Z}({\rm M_0})$  for each nonabelian maximal subgroup, and G  $\epsilon$  3 .

Proposition 3.3: If G  $\epsilon$  3 and if G has both a Redei maximal subgroup and an abelian maximal subgroup, then for each Redei subgroup R of G, R is maximal in G and Z(G) = Z(R).

Proof: Let  $R_0$  be the Redei maximal subgroup and let  $M_0$  be the abelian maximal subgroup. Then  $G=M_0R_0$ .  $M_0\cap R_0$  is, therefore, a maximal subgroup of  $R_0$ . Since  $Z(R_0) < M_0 \cap R_0 < M_0$ , it follows that  $Z(R_0) \le Z(G)$ . However,  $Z(G) \le Z(R_0)$  since  $G \in \mathcal{J}$ . Thus  $Z(G) = Z(R_0)$ . But  $(R_0 : Z(R_0)) = \mathbf{p}^2$  since  $R_0$  is Redei. This implies that Z(G) has index  $\mathbf{p}^3$  in G. Consequently, Z(G) = Z(R) for each Redei subgroup R, and R is maximal in G.

The groups from  $\mathcal{J}$  with a Redei maximal subgroup can be separated into two types:

- (i) groups with each Redei subgroup as a maximal subgroup, or
- (ii) groups with a Redei subgroup that is not maximal.

  Because of Proposition 3.3, no maximal subgroup of a group of type (ii) is abelian.

<u>Proposition 3.4</u>: Let  $G \in \mathcal{J}$ . Let G have a Redei maximal subgroup  $R_0$  and a maximal subgroup that is nonabelian and non-Redei. Then, each nonabelian, non-Redei subgroup N of G has exactly one abelian maximal subgroup, and all Redei subgroups of N have the same index in N. Furthermore,  $(Z(R)\colon Z(G)) \leq p$  for each Redei subgroup R of G different from  $R_0$ .

Proof: Let N be a nonabelian, non-Redei proper subgroup of G. Let R be a Redei subgroup of N. Since  $G=R_0N$ , then  $R_0\cap N$  is a maximal abelian subgroup of N. Because class  $\mathcal J$  is subgroup inherited, N  $\varepsilon$   $\mathcal J$ . However, by Proposition 3.1  $R_0\cap N$  is the only abelian maximal subgroup of N.

Let R be a Redei subgroup of N . Then N =  $(R_0 \cap N)R$ , which implies that  $R_0 \cap R$  is a maximal subgroup of R . Therefore,  $Z(R) < \Phi(R) < R_0 \cap R \le R_0 \cap N$  . Hence,  $Z(R) \le Z(N)$  . But from (1.1)  $Z(N) \le Z(R)$  . Hence, Z(N) = Z(R) . Now  $(R: Z(R)) = p^2$  for each Redei subgroup R of N from which it follows that all Redei subgroups of N have the same index in N .

To prove the index relation for Redei subgroups, let  $R^*$  be any Redei subgroup of G and let  $N^*$  be a maximal subgroup of G containing  $R^*$ . By the first part of the proof,  $Z(N^*) = Z(R^*)$ . Thus  $N^* \leq C_G(Z(R^*))$ . If  $C_G(Z(R^*)) = G$ , then  $Z(R^*) = Z(G)$ . On the other hand, let  $N^* = C_G(Z(R^*))$ . Since  $N^* \cap R_0$  is a maximal abelian subgroup of both  $N^*$  and  $R_0$ , then  $Z(R_0)Z(R^*) \leq N^* \cap R_0$ . But  $Z(R^*) \not \leq Z(R_0)$ , which forces the equality  $Z(R_0)Z(R^*) = N^* \cap R_0$ . Hence,  $(Z(R^*): Z(R_0) \cap Z(R^*)) = p$ . However,  $Z(R_0) \cap Z(R^*) = Z(G)$ . This leads to the conclusion that  $(Z(R): Z(G)) \leq p$  for each Redei subgroup R of G different from  $R_0$ .

The above theorem shows that if a group from class 3 with a Redei maximal subgroup is of type (ii), then no proper nonabelian, non-Redei subgroup of G can belong to either class 3 or class R. Therefore, any group in  $3 \sim 3$  has only the groups of type (i), with an abelian maximal subgroup, as minimal non-Redei subgroups.

THEOREM 3.1: If  $G \in \mathcal{J}$  and if G has both an abelian maximal subgroup and a Redei maximal subgroup, then G is one of the following

nonmetacyclic groups.

- (1)  $G = R_0 < c | c^{3^2} = 1 > \text{ where } R_0 = < a, b | a^{3^2} = b^3 = 1, [a, b] = a^3 > ,$   $[c,b] = 1, [a,c] = b \text{ and } c^3 = a^{3u} \text{ where } u \neq 0 \pmod{3}. \qquad |G| = 3^4.$
- (2)  $G = M_0 < b | b^p = 1 > where <math>M_0 = < a_1 | a_1^p = 1 > w < a_2 | a_2^p = 1 > w < a_3 | a_3^p = 1 >$ ,  $[a_1,b] = a_2$ ,  $[a_2,b] = a_3$ .  $|G| = p^4$ .
- (3)  $G = M_0 < c | c^3 = 1 > where M_0 = < a,b | a^3 = b^3 = 1, [a,b] = 1 > ,$   $[c,a] = b \text{ and } [c,b] = a^6.$   $[G| = 3^4.$
- (4)  $G = R_0 < c | c^3 = 1 > \text{where } R_0 = < a, b | a^3 = b^3 = 1, [a, b] = a^3 > ,$  [c,a] = b and [b,c] = 1.  $|G| = 3^4.$
- (6)  $G = R_0 < c | c^p = 1 > \text{ where } R_0 = < a, b | a^{p^m} = b^p = 1, [a, b] = a^{p^{m-1}} > ,$  [a,c] = b and [b,c] = 1.  $|G| = p^{m+2} \ge p^4.$
- (7)  $G = M_0 < c | c^p = 1 > \text{where } M_0 = < a, b | ap^2 = b^p = 1, [a,b] = 1 >$ , [a,c] = b and [b,c] =  $a^{rp}$  where r is either 1 or a quadratic non-residue for p > 3 and r = 1 for p = 3.  $|G| = p^{4}$ .
- (8)  $G = R_0 < d | d^{p^2} = 1$  >where  $R_0 = < a, b | a^{p^m} = b^p = c^p = 1$ , c = [a, b], [a, c] = [b, c] = 1 >, [a, d] = b, [b, d] = 1 and  $c = d^{p}$  for  $r \neq 0 \pmod{p}$ .  $|G| = p^{m+3} \geq p^5$ .
- (9)  $G = R_0 < c | c^p = 1 > \text{where } R_0 = < a, b | ap^m = b^p = 1, [a, b] = ap^{m-1} > ,$   $[b,c] = 1 \text{ and } [a,c] = a^p . \qquad |G| = p^{m+2} \ge p^4 .$

- (10)  $G = R_0 < c | c^p = 1 > \text{ where } R_0 = < a, b | ap^m = bp^2 = 1, [a, b] = ap^{m-1} >$ ,  $[a, c] = b^p \text{ and } [c, b] = 1$ .  $|G| = p^{m+3} \ge p^5$ .
- (11)  $G = R_0 < d | d^p^2 = 1 > where R_0 = < a, b | a^p = b^p^2 = c^p = 1, c = [a, b], [a, c] = [b, c] = 1 > ,$   $d^p = c$ , [d, b] = 1 and  $[d, a] = c^r z^{sp}$  for r = 0 or 1 and  $s \not\equiv 0 \pmod p$ .  $|G| = p^{m+4} \ge p^6.$

Each of the next seven lemmas constitutes a part of the proof of Theorem 3.1. The basic hypotheses are the same for all of these lemmas. For convenience these hypotheses are summarized as follows.

(3.1) G  $\epsilon$   $\emph{3}$  . G has a Redei maximal subgroup  $\,{\rm R}_0$  , an abelian maximal subgroup  $\,{\rm M}_0$  , and a nonmetacyclic maximal subgroup.

Lemma 3.1: Given hypotheses (3.1). Then the index of  $\Phi(G)$  in G is either  $p^2$  or  $p^3$ . In the case where  $(G:\Phi(G))=p^2$ , then both  $|G'|=p^2$  and all the nonabelian subgroups have the same commutator subgroup of order p. When  $(G:\Phi(G))=p^3$ , then G is a regular group.

Proof: Since  $R_0$  is maximal in G,  $(G: \Phi(R_0)) = p^3$ . From  $\Phi(R_0) \le \Phi(G)$  it follows that  $(G: \Phi(G)) \le p^3$ . Since G is not cyclic,  $(G: \Phi(G)) \ge p^2$ . Thus,  $p^2 \le (G: \Phi(G)) \le p^3$ .

Suppose that  $(G: \Phi(G)) = p^2$ . To show the commutators are the same, consider  $G/R_0^i$ .  $G/R_0^i$  has two abelian maximal subgroups  $R_0/R_0^i$  and  $M_0/R_0^i$ , which implies that  $Z(G/R_0^i) = \Phi(G/R_0^i)$ . Then each maximal subgroup of  $G/R_0^i$  is abelian. Since  $|R_0^i| = p$ ,  $R_0^i$  is the commutator subgroup of each nonabelian maximal subgroup of G.

If  $G/R_0'$  is abelian, then  $G'=R_0'\leq Z(R_0)$ . Thus, by Proposition 3.3,  $G'\leq Z(G)$ . By Proposition 1.5,  $[x^p,y]=[x,y]^p=1$  for each pair of elements x,y from G. Therefore,  $\upsilon_1(G)\leq Z(G)$ . From  $\varphi(G)=G'\upsilon_1(G)$ , it now follows that  $\varphi(G)\leq Z(G)$ . This is a contradiction to Proposition 3.3 by which  $Z(G)=\varphi(R_0)<\varphi(G)$ . Hence,  $G/R_0'$  is a Redei group, which leads to  $|G'|=p^2$ .

If  $(G: \Phi(G)) = p^3$ , then  $\Phi(G) = Z(G)$ . Since  $G' \le \Phi(G) = Z(G)$ , the class of G is 2. By Theorem 1.7(a), G is regular.

Lemma 3.2: Given hypotheses (3.1), then  $p^3 \le |\Omega_1(G)| \le p^4$ .

Proof: By hypothesis, there is a nonmetacyclic maximal subgroup  $N_0$ . By Theorem 1.9,  $|N_0/v_1(N_0)| \geq p^3$ . Since  $N_0$  is either Redei or abelian, then  $N_0$  is regular. Consequently, by Theorem 1.6,  $|\Omega_1(N_0)| \geq p^3$ . But  $\Omega_1(N_0) \leq \Omega_1(G)$  so that  $|\Omega_1(G)| \geq p^3$ .

For the other inequality in the conclusion, two cases are derived from Lemma 3.1, which states  $p^2 \leq (G: \Phi(G)) \leq p^3$ . First, if  $(G: \Phi(G)) = p^3$ , then G is regular. By Theorem 1.8,  $(N: \upsilon_1(N)) \leq p^3$  for each nonabelian maximal subgroup. Thus,  $(G: \upsilon_1(N)) \leq p^4$  for each nonabelian maximal subgroup. Since  $\upsilon_1(N) \leq \upsilon_1(G)$ , then  $(G: \upsilon_1(G)) \leq p^4$ . From regularity, it follows that  $|\Omega_1(G)| \leq p^4$ . On the other hand, if  $(G: \Phi(G)) = p^2$ , then by Lemma 2.1,  $|G'| = p^2$ . By Proposition 1.6,  $\Phi(G')G_3$  is the only maximal subgroup of G' which is normal in G. Thus  $\Phi(G')G_3 = R_0'$ . But, Proposition 1.7 implies that  $G/R_0'$  is a nonmetacyclic Redei group. This in turn implies that  $(G/R_0'; \upsilon_1(G/R_0')) = p^3$  by Theorem 1.8. From the regularity of Redei groups,  $|\Omega_1(G/R_0')| = p^3$ .

Now  $R_0' \leq \Omega_1(G)$ . Thus  $\Omega_1(G)/R_0' \leq \Omega_1(G/R_0')$ , from which it can be concluded that  $|\Omega_1(G)| \leq p^4$ .

Hence,  $p^3 \le |\Omega_1(G)| \le p^4$ .

Lemma 3.3: Assume hypotheses (3.1). If  $(G: \Phi(G)) = p^2$  and if  $|\Omega_1(G)| = p^4$ , then either  $G = \Omega_1(G)$  or  $\Phi(G)\Omega_1(G)$  is a maximal subgroup. If  $G = \Omega_1(G)$ , then G is the group of type (2), type (3), or type (4). If  $G \neq \Omega_1(G)$ , then G is the group of type (5).

Proof: If  $\Omega_1(G) \leq \Phi(G)$ , then  $\Omega_1(G) \leq R_0$ . But then,  $\Omega_1(G) = \Omega_1(R_0)$ . However,  $R_0$  is regular which, from Theorems 1.8 and 1.6, implies that  $\left[\Omega_1(R_0)\right] \leq p^3$ ; a contradiction is reached. Hence, either  $G = \Omega_1(G)$  or  $\Phi(G)\Omega_1(G)$  is a maximal subgroup of G.

Case 1:  $G = \Omega_1(G)$ . By Lemma 3.1 and Proposition 3.3,  $\Phi(G) = G'$  and |Z(G)| = p. Thus,  $G_3 = [G,G'] = Z(G)$  and the class of G is 3.

If p>3 , then G is regular by Theorem 1.7(a). This implies that  $\upsilon_1(G)=1$  by Theorem 1.6. Thus, the abelian maximal subgroup  $\,M_0^{}$  has the form

$$\mathbb{M}_0 = \langle \ \mathbf{a_1} | \mathbf{a_1^p} = 1 \rangle \otimes \langle \ \mathbf{a_2} | \mathbf{a_2^p} = 1 \rangle \otimes \langle \ \mathbf{a_3} | \mathbf{a_3^p} = 1 \rangle \; .$$

It may be assumed that  $G' = \langle a_2, a_3 \rangle$  and that  $Z(G) = \langle a_3 \rangle$ . Let b be any element in  $G \sim M_0$ . Then,

$$G = M_0 < b|b^p = 1 > , [a_1,b] = a_2^r a_3^s , \text{ and } [a_2,b] = a_3^t .$$

From Lemma 3.1, G/Z(G) is nonabelian, which implies that  $r \not\equiv 0 \pmod p$ . Since  $a_2 \not\in Z(G)$  ,  $t \not\equiv 0 \pmod p$  . Let

$$\overline{a}_2 = a_2^{rs}$$
 and  $\overline{a}_3 = a_3^{rt}$ .

Then,  $G' = \langle \overline{a}_2, \overline{a}_3 \rangle$  and  $[a_1,b] = \overline{a}_2$ . Also, by Proposition 1.5,  $[\overline{a}_2,b] = \overline{a}_3$ . Thus,

$$G = M_0 < b | b^p = 1 > \text{where } M_0 = < a_1 | a_1^p = 1 > \otimes < \overline{a}_2 | \overline{a}_2^p = 1 > \otimes < \overline{a}_3 | \overline{a}_3^p = 1 > ,$$
 
$$[a_1, b] = \overline{a}_2 \quad \text{and} \quad [\overline{a}_2, b] = \overline{a}_3 . \qquad \cdot$$

This is the group of type (2) in Theorem 3.1.

Now consider p=3. The normetacyclic maximal subgroup  $N_0$ , which is either Redei or abelian, has order  $3^3$ . Then,  $\upsilon_1(N_0)=1$  by Theorem 1.9. This implies that  $\Omega_1(N_0)=N_0$ . But G', which has order  $p^2$ , is contained in  $N_0$  so that G' is elementary abelian. If  $\upsilon_1(G)=1$ , then G automatically satisfies the definition of regularity. However, by Theorem 1.7(a), G is not a regular group. Hence,  $\upsilon_1(G)\neq 1$ . Since G is not metacyclic, it then follows that  $\upsilon_1(G)=Z(G)$ . This implies that G must have a metacyclic maximal subgroup.

By Proposition 1.4(a), the abelian maximal subgroup  $\rm\,M_{0}$  is the G-centralizer of  $\rm\,G^{1}$  .  $\rm\,M_{0}$  is either metacyclic or nonmetacyclic.

Suppose first that  ${\rm M}_{\rm O}$  is metacyclic. Thus,

$$M_0 = \langle a,b | a^{3^2} = b^3 = 1$$
, [a,b] = 1 > .

Since G' is elementary abelian, then

$$G' = \Omega_1(M_0) = \langle a^3 \rangle \otimes \langle b \rangle$$
 and  $Z(G) = \langle a^3 \rangle$ .

There exists an element c in  $G \sim M_0$  such that  $c^3 = 1$  (otherwise,  $\Omega_1(G) \leq M_0$ ). Hence,  $G = M_0 < c | c^3 = 1 > .$  Now G' < c > is a non-metacyclic group; denote G' < c > by  $N_0$ . If [c,b] = 1, then  $Z(G) = \langle a^3 \rangle \otimes \langle b \rangle$ , which is a contradiction to |Z(G)| = p. Hence,  $[c,b] \neq 1$ . But  $[c,b] \in [G,G'] = Z(G)$ , which implies that

$$[c,b] = a^{\alpha\beta}$$
 where  $\alpha \not\equiv 0 \pmod{3}$ .

If [c,a]  $\epsilon$  Z(G), then it follows that G' = Z(G), which is a contradiction to |G'| = p<sup>2</sup>. Consequently, [c,a]  $\epsilon$  G'  $\sim$  < a<sup>3</sup> >, so that

[c,a] = 
$$a^{\gamma 3}b^{\beta}$$
 where  $\beta \neq 0 \pmod{3}$ .

There exists a  $\delta$  such that  $\beta\delta\equiv 1 \pmod 3$ . Let  $a_1=a^\delta$  and  $b_1=a^{\gamma\delta 3}b$ . By Proposition 1.5,  $[c,a_1]=b_1$ ,  $[a_1,b]=1$ , and  $[c,b_1]=a^{\alpha\beta 3}=a^{\alpha\beta^2 3}$ . It is possible, then, to drop the subscripts, so the group is defined by

$$G = M_0 < c | c^3 = 1 > \text{where } M_0 = < a, b | a^3 = b^3 = [a, b] = 1 > ,$$
(3.2)

[c,a] = b and  $[c,b] = a^{r3}$  for  $r \neq 0 \pmod{3}$ .

The relations in (3.2) lead to

$$c^{y}b^{x} = a^{xy}b^{x}c^{y}$$
 and  $c^{y}a^{x} = a^{x} + y(y+1)x^{2}$ 

which in turn give

(3.3) 
$$d^3 = a^{[1 + t(t+1)r/2 + 2t(2t+1)r/2]3q}$$
 where  $d = a^q b^s c^t$ .

Suppose that  $d^3 = 1$ . Then,

$$[1 + (5t^2 + 3t)r/2]q \equiv 0 \pmod{3}$$
.

If r=1 and  $q \not\equiv 0 \pmod 3$ , then  $1+(5t^2+3t)/2 \equiv 0 \pmod 3$ . It follows that  $t^2 \equiv 2 \pmod 3$ , which has no solution. If r=1 and  $q \equiv 0 \pmod 3$ , then  $d \in \mathbb{N}_0 = G' < c >$ , which implies that  $\Omega_1(G) = \mathbb{N}_0$ . This is a contradiction to  $G = \Omega_1(G)$ . For r=2, consider the element ac which is not an element of  $\mathbb{N}_0$ . By (3.3), (ac) $^3=1$ . Consequently,  $\Omega_1(G) = G$ . Hence, in defining relations (3.2), r=2. This group is the group of type (3) in Theorem 3.1.

For the other possibility where  $\,{\rm M}_{0}\,$  is not a metacyclic group, it follows that G has a metacyclic maximal subgroup  $\,R_{1}$  . Let

$$R_{\gamma} = \langle a,b | a^{3^2} = b^3 = 1$$
, [a,b] =  $a^3 > .$ 

Since G' is elementary abelian, then

$$G' = \langle a^3 \rangle \otimes \langle b \rangle \text{ and } Z(G) = \langle a^3 \rangle$$
.

There exists an element  $c \in M_0 \sim N_0$  such that  $c^3 = 1$ . Then [b,c]=1. If  $[c,a] \in Z(G)$ , then G'=Z(G) which is a contradiction to  $|G'|=p^2$ . Thus,

[b,c] = 1 and  $[c,a] = b^s a^{r3}$  where  $s \neq 0 \pmod{3}$ .

There exists a t such that st  $\equiv$  1(mod 3). Since [[c,a], c] = 1, Proposition 1.5 implies that [c<sup>t</sup>,a] = ba<sup>rt3</sup>. Let c<sub>1</sub> = c<sup>t</sup> and b<sub>1</sub> = ba<sup>rt3</sup>. Then, it is possible to drop the subscripts to get the defining relations:

$$G = R_1 < c | c^3 = 1 > \text{ where } R_1 = < a, b | a^3 = b^3 = 1, [a, b] = a^3 > [c, a] = b \text{ and } [b, c] = 1.$$

This is the group of type (4) in Theorem 3.1.

Since  $(M_0\colon \Phi(G))=p$ , then  $\Phi(G)\cap \Omega_1(G)$  is a maximal subgroup of  $\Omega_1(G)$  and has order  $p^3$ . Moreover,  $\Omega_1(G)$  is abelian which implies that  $\Phi(G)\cap \Omega_1(G)=\Omega_1(\Phi(G))$ . In view of the fact that  $\Omega_1(\Phi(G))\leq M$  for each maximal subgroup M of G, it follows that  $\Omega_1(\Phi(G))\leq \Omega_1(M)$ . Thus, each nonabelian maximal subgroup is nonmetacyclic.

By Lemma 3.1,  $G/R_0^1$  is a Redei group. From Propositions 1.6

and 1.7,

$$G/R_0' = \langle \overline{a}, \overline{b} | \overline{a}^{p^m} = \overline{b}^{p^m} = \overline{c}^p = 1, \overline{c} = [\overline{a}, \overline{b}], [\overline{a}, \overline{c}] = [\overline{b}, \overline{c}] = 1 \rangle .$$

If  $m, n \geq 2$ , then  $\Omega_1(G/R_0') = \langle \overline{a}^{p^{m-1}} \rangle \otimes \langle \overline{b}^{p^{m-1}} \rangle \otimes \langle \overline{c} \rangle \leq \phi(G)/R_0'$ . However,  $\Omega_1(G)/R_0' \leq \Omega_1(G/A)$  so that  $\Omega_1(G) \leq \phi(G)$ . This is a contradiction. Thus, one of m,n must be 1; say n=1. There exists an element  $b \in \Omega_1(G) \sim \phi(G)$  such that  $bR_0' = \overline{b}$ ;  $b^p = 1$ . Now  $\overline{a} \in M/R_0'$  for some maximal subgroup M of G. There exists an element  $a \in M$  such that  $aR_0' = \overline{a}$ . Then  $ap^m \in R_0'$ . If  $M = M_0$ , then  $G/R_0' = \langle \overline{a}, \overline{b} \rangle = M_0/R_0'$ , which is a contradiction. Thus,  $M \neq M_0$ . Since M is normetacyclic and Redei,  $M' \cap \psi_1(M) = 1$ . However,  $M' = R_0'$  so that  $ap^m = 1$ . Let c = [a,b]. Then  $cR_0' = \overline{c}$ . Both b and c are elements in  $M_0$  from which it follows that [b,c] = 1. Furthermore,  $[a,c] \neq 1$  (otherwise  $G' = R_0'$ ). Let d = [a,c]. Hence,

$$G = R_1 < b | b^p = 1 > \text{ where}$$

$$R_1 = \langle a,c | a^{p^m} = c^p = d^p = 1, d = [a,c], [a,d] = [c,d] = 1 \rangle$$

$$[a,b] = c$$
 and  $[b,c] = [b,d] = 1$ .

This is the group of type (5) in Theorem 3.1.

Lemma 3.4: Given hypotheses (3.1). If (G:  $\phi(G)$ ) =  $p^2$  and if  $|\Omega_1(G)| = p^3$ , then either  $\phi(G)\Omega_1(G)$  is a maximal subgroup or

 $\Omega_1(G) \leq \Phi(G)$  . If  $\Omega_1(G) \leq \Phi(G)$  , then G is the group of type (8). If  $\Omega_1(G) \not\leq \Phi(G)$  , then G is the group of type (6) or type (7).

Proof: For the normetacyclic maximal subgroup  $N_0$ ,  $\left|\Omega_1(N_0)\right|=p^3=\left|\Omega_1(G)\right|$ . This implies that  $\Omega_1(G)< G$ . Thus, either  $\Omega_1(G)<\Phi(G)$  or  $\Phi(G)\Omega_1(G)$  is maximal.

Let R be a nonabelian maximal subgroup different from N $_0$ . Then R is metacyclic and  $\upsilon_1(R)=\phi(R)$  where  $(R:\upsilon_1(R))=p^2$ . G, however, is not metacyclic, which implies that  $(G:\upsilon_1(G))\geq p^3$ . From  $\upsilon_1(R)\leq \upsilon_1(G)$ , it follows that  $\upsilon_1(G)=\upsilon_1(R)=Z(R)$ . In particular,

(3.4) 
$$v_1(G) = Z(G)$$
.

Suppose first of all that  $|G| > p^4$ . If  $N_0$  is nonabelian, then  $N_0' \cap \upsilon_1(N_0) = 1$  by Theorem 1.8, and  $|N_0' \Omega_1(\upsilon_1(N_0))| \ge p^2$ . However,  $|N_0' \Omega_1(\upsilon_1(N_0))| \le \Phi(N_0) \le \Phi(G)$ , so that  $|N_0' \Omega_1(\upsilon_1(N_0))| \le \Omega_1(\Phi(G))$ . But  $|\Omega_1(\Phi(G))| = p^2$  since  $\Omega_1(G) \ne \Phi(G)$ . Thus,  $|N_0' \Omega_1(\upsilon_1(N_0))| \le \Omega_1(\Phi(G))$  is an elementary abelian subgroup of order  $|p|^2$  which is contained in  $|\Phi(G)|$ . Therefore,  $|N_0' \Omega_1(\upsilon_1(N_0))| = \Omega_1(\Phi(G))$ . Since  $|\Omega_1(M)| \le \Omega_1(\Phi(G))$  for a metacyclic maximal subgroup M , it follows that  $|\Omega_1(M)| = N_0' \Omega_1(\upsilon_1(N_0))$ .

Since N<sub>0</sub> is nonabelian,  $\Phi(N_0) = Z(N_0) = Z(G)$ . Consequently,  $\Phi(N_0) = \upsilon_1(G)$ . Now,  $(\Phi(G) : \upsilon_1(G)) = p$ , which implies that  $G' \not = \Phi(N_0)$ . Furthermore, since  $G' \leq R$  and since  $\Omega_1(R) = N_0'\Omega_1(\upsilon_1(N_0)) \leq \Phi(N_0) = \upsilon_1(G)$ , then  $G' \not = \Omega_1(R)$ . Therefore, G' is cyclic. Let  $G' = \langle c | c^p^2 = 1 \rangle$ . Then  $c^p \in \Omega_1(G')$ . By Lemma 3.1,  $N_0' = \Omega_1(G')$ . Since  $G' \leq N_0$ ,  $c^p \in \upsilon_1(N_0)$ . Thus,  $N_0' \cap \upsilon_1(N_0) \neq 1$ , which contradicts the fact that  $N_0$  is nonmetacyclic. Hence,  $N_0$  must be abelian, that is,  $N_0 = M_0$ .

By (3.4),  $\upsilon_1(G)$  has index  $p^3$  in G. This implies that  $\upsilon_1(G) = \upsilon_1(M)$  for each metacyclic maximal subgroup M, whereby  $\upsilon_1(M_0) \leq \upsilon_1(M)$ . Moreover,  $G' \not\leq \upsilon_1(G)$  from which it can be concluded that  $G' \not\leq \upsilon_1(M_0)$ . If  $G' \cap \upsilon_1(M_0) = 1$ , then G', as a subgroup of  $M_0$ , must be elementary abelian of order  $p^2$ . Then,  $G' \Omega_1(\upsilon_1(M_0)) \leq \Phi(G)$ . But  $|G' \Omega_1(\upsilon_1(M_0))| \geq p^3$  from which it follows that  $\Omega_1(M_0) = G' \Omega_1(\upsilon_1(M_0)) \leq \Phi(G)$ . This is a contradiction to  $N_0 = M_0 = \Phi(G)\Omega_1(G)$ . Thus,  $G' \cap \upsilon_1(M_0) \neq 1$ . By Lemma 3.1,  $G' \cap \upsilon_1(M_0) = R_0'$ .

Consider now  $G/R_0'$ . Denote this by  $\overline{G}$ . Since  $\Omega_1(G) \not \leq \Phi(G)$ , then  $\Omega_1(\overline{G}) \leq \Phi(\overline{G})$ . From Lemma 3.1 and Propositions 1.6 and 1.7,  $\overline{G}$  is of the form  $\langle \overline{a}, \overline{b} | \overline{a}^{p^m} = \overline{b^p} = \overline{c}^p = 1$ ,  $\overline{c} = [\overline{a}, \overline{b}]$ ,  $[\overline{a}, \overline{c}] = [\overline{b}, \overline{c}] = 1 \rangle$ . Let a and b be elements of G such that  $aR_0' = \overline{a}$  and  $bR_0' = \overline{b}$ . Let c = [a,b]. Then  $cR_0' = \overline{c}$  and  $c^p \in R_0'$ . In addition,  $\overline{M}_0 = \overline{\Phi(G)\Omega_1(G)} \leq \Phi(\overline{G})\Omega_1(\overline{G})$ . If  $\overline{M} < \Phi(\overline{G})\Omega_1(\overline{G})$ , then  $\overline{G} = \Omega_1(\overline{G})$ . Since  $\overline{G}$  is Redei and normetacyclic, it follows that  $\upsilon_1(\overline{G}) = 1$ . Thus, m = 1 and  $|\overline{G}| = p^3$ . This is a contradiction to  $|G| > p^4$ . Hence,  $\overline{M}_0 = \Phi(\overline{G})\Omega_1(\overline{G})$ . Since  $\overline{b} \in \overline{M}_0$ , then  $b \in M_0$ . By Proposition 1.5, this leads to  $c^p = [a,b]^p = [a,b^p]$ . But,  $b^p \in \upsilon_1(G) = Z(G)$  which implies that  $[a,b^p] = 1$ . In particular,  $c^p = 1$ . Consequently, G' is elementary abelian. From  $|\Omega_1(\Phi(G))| = p^2$  and from  $G' \not \prec \upsilon_1(G)$ , it follows

that  $|\Omega_1(\upsilon_1(G))| = p$ . This, in turn, implies that  $\upsilon_1(G)$  is cyclic. It is now possible to describe the metacyclic maximal subgroups of G. Let M be such a subgroup. Then,

$$M = \langle a, b | a^{p^m} = b^p = 1, [a, b] = a^{p^{m-1}} \rangle$$
.

There exists an element  $c \in M_0 \sim M$  such that  $c^p = 1$ . Then,

$$G = M < c | c^{p} = 1 > .$$

Both [b,c] and [a,c] belong to G'. Also, b  $\epsilon \Omega_1(M) \leq \Phi(G)$ , so that b  $\epsilon \Omega_1(\Phi(G)) = G$ '. But then, b  $\epsilon M_0$ . Hence,

$$[b,c] = 1$$
 and  $[a,c] = a^{rp} b^{s}$ ,  $s \neq 0 \pmod{p}$ .

Let  $c_1=c^t$  where t is such that ts  $\equiv 1 \pmod p$ . Since  $[a,c] \in M_0$ , then [[a,c],c]=1. By Proposition 1.5,  $[a,c]^t=[a,c^t]$ . Thus,  $[a,c_1]=a^{trp^{m-1}}b$ .

Let  $b_1 = a^{trp^{m-1}}b$ , then  $b_1^p = 1$  and  $b_1 \not\in Z(G)$  (otherwise,  $b \in Z(G)$ ). Now  $[a,b_1] = [a,a^{trp^{m-1}}b] = a^{p^{m-1}}$  since  $a^{trp^{m-1}}e$   $Z(G) = v_1(G)$ . Also  $[b_1,c_1] = 1$  and  $b_1 \not\in \Phi(M)$ . Thus,

$$M = \langle a, b_1 | a^{p^m} = b_1^p = 1, [a, b_1] = a^{p^{m-1}} \rangle$$
 and  $G = M \langle c_1 | c_1^p = 1 \rangle$   
where  $[a, c_1] = b_1$  and  $[b_1, c_1] = 1$ .

This is the group of type (6) in Theorem 3.1.

Now suppose that  $|G|=p^4$ . Then,  $N_0=\Omega_1(N_0)=\Omega_1(G)$ ,  $Z(G)=R_0'=\upsilon_1(G)$  with order p, and G' has order  $p^2$ . Since  $N_0$ 

is regular (either abelian or Redei), then  $\upsilon_1(N_0)=1$ . It follows from  $G'\leq N_0$  that G' is elementary abelian of order  $p^2$ .

Suppose that  $N_0$  is abelian. There exists then a nonabelian metacyclic maximal subgroup M with defining relations

$$M = \langle a, b | a^2 = b^p = 1, [a, b] = a^p \rangle$$
.

There exists an element  $\ c \ \epsilon \ N_0 \sim M$  . By the argument that has been used for  $|G| > p^{ij}$  , it follows that

$$G = M < c | c^p = 1 > where [a,c] = b$$
 and  $[b,c] = 1$ ,

which is the group of type (6) in Theorem 3.1.

If, on the other hand,  $\,{\rm N}_0\,$  is nonabelian, then the abelian maximal subgroup  $\,{\rm M}_0\,$  is metacyclic and has defining relations

$$M_0 = \langle a | a^{p^2} = 1 \rangle \otimes \langle b | b^p = 1 \rangle$$
.

Then  $G' = \langle a^p \rangle \otimes \langle b \rangle$  and  $Z(G) = R_0' = \upsilon_1(G) = \langle a^p \rangle$ . There exists an element  $c \in N_0 \sim M_0$  and

$$G = [M_0] < c | c^p = 1 > .$$

Since both b and c belong to N<sub>0</sub>, [b,c]  $\epsilon$  N<sub>0</sub> = R<sub>0</sub> =  $\upsilon_1(G)$  = <  $a^p$  > . Also, [a,c]  $\epsilon$  G'  $\sim$  Z(G) . Thus,

$$[b,c] = a^{rp}$$
,  $r \not\equiv 0 \pmod{p}$  and  $[a,c] = a^{sp}b^t$ ,  $t \not\equiv 0 \pmod{p}$ .

Let 
$$b_1 = a^{sp}b^t$$
. Then,  $b_1^p = 1$  and  $b_1 \in M_0 \sim v_1(M_0)$ . Then  $M_0 = \langle a|a^{p^2} = 1 \rangle \otimes \langle b_1|b_1^p = 1 \rangle$ .  $[b_1,c] = [a^{sp}b^t,c] = [b^t,c]$ . Now

[b,c]  $\in \mathbb{N}_0^! = \mathbb{Z}(\mathbb{N}_0)$  so by Lemma 1.3,  $[b^t,c] = [b,c]^t$ . Denote tr by u. Then  $u \neq 0 \pmod{p}$  and  $[b_1,c] = a^{up}$ .

It is now possible to drop the subscripts to get  $G=M_0< c|c^p=1>$  where  $M_0=< a,b|a^p^2=b^p=1$ , [a,b]=1>,  $[b,c]=a^{up}$  where  $u\not\equiv 0 \pmod p$  and [a,c]=b. These relations give rise to  $[b,c^\alpha]=a^{\alpha up}$  and  $[a,c^\alpha]=a^{\alpha(\alpha-1)p/2}b^\alpha$ . For  $\alpha\not\equiv 0 \pmod 3$ ,  $G=M_0< c^\alpha>$ . Thus, for  $c_1=c^\alpha$ , and  $b_1=a^{\alpha(\alpha-1)p/2}b^\alpha$ , then

(3.5) 
$$G = M_0 < c_1 > \text{where } M_0 = < a, b_1 | a^{p^2} = b_1^p = 1, [a, b_1] = 1 > ,$$

$$[a, c_1] = b_1 \text{ and } [b_1, c_1] = a^{\alpha^2 up}.$$

Now  $u \not\equiv 0 \pmod p$  so there exists v such that  $vu \equiv 1 \pmod p$ . If u is a quadratic residue, then v is a quadratic residue, that is, there exists a k such that  $k^2u \equiv 1$ . If u is not a quadratic residue and t is not a quadratic residue, then from tvu = t it follows that tv is a quadratic residue. Then there is a k such that  $k^2u = t$ . Thus for  $s \in \{1,t\}$  where t is a quadratic nonresidue, there exists a k such that  $k^2u = s$ . Hence, it can be assumed, without loss of generality, that  $\alpha^2u$  is either 1 or a quadratic nonresidue.

Since Z(G) < G', G has class 3. If p > 3, then from Proposition 1.7(a), G is regular; it follows that a group will exist for each value of r. However, if p = 3, then the proof of Lemma 3.3 has shown that  $\Omega_1(G) = G$  when the quadratic nonresidue  $\alpha^2 u$  is equal to 2. Hence, for p = 3,  $\alpha^2 u$  can only be 1.

The groups satisfying (3.5) are the groups of type (7) in Theorem 3.1.

Case 2:  $\Omega_1(G) \leq \Phi(G)$ . Then,  $\Omega_1(M) = \Omega_1(G)$  for each maximal subgroup M , whereby each nonabelian maximal subgroup is nonmetacyclic. By Proposition 1.6,  $R_0'$  is the only maximal subgroup of G' which is normal in G. Since  $M' \cap \upsilon_1(M) = 1$  for each nonabelian maximal subgroup M , it follows that  $G' \cap \upsilon_1(M) = 1$ . Now by Lemma 3.2,  $|G'| = p^2$ . If G' is cyclic, then  $M' \leq \upsilon_1(M)$  which is a contradiction to  $G' \cap \upsilon_1(M) = 1$ . Thus, G' is elementary abelian. This implies that  $\upsilon_1(M)$  is cyclic. Consequently, for the nonabelian Redei subgroup  $R_0$ ,

$$R_0 = \langle a, b | a^{p^m} = b^p = c^p = 1$$
,  $c = [a,b]$ ,  $[b,c] = [a,c] = 1 \rangle$ .

To determine the relation between G' and  $\upsilon_1(G)$  suppose that  $G'\cap\upsilon_1(G)=1\ .\ \ \text{Since}\ \ (\Phi(G)\colon\Phi(M))=p\ \ \text{for each nonabelian maximal}$  subgroup M , then  $\upsilon_1(G)=\upsilon_1(M);\ \text{so}\ \upsilon_1(G)\ \ \text{is cyclic}.$  Let  $G/G'=\langle\ \overline{x}>\otimes<\overline{y}>\ \text{where}\ \ \overline{x}^{p^\alpha}=1\ \ \text{and}\ \ \overline{y}^{p^\beta}=1\ .$  Let x and y be elements from G such that  $xG'=\overline{x}$  and  $yG'=\overline{y}$  . Then  $x^{p^\alpha}\in G'$  and  $y^{p^\beta}\in G'$  where  $\alpha,\,\beta\geq 1$  . Since  $G'\cap\upsilon_1(G)=1$  ,  $x^{p^\alpha}=y^{p^\beta}=1$  . However, since  $\Omega_1(G)\leq \Phi(G)$  , then  $\alpha,\,\beta\geq 2$  . Thus,  $\Phi(G/G')=\langle\ \overline{x}^p>\otimes<\overline{y}^p>$  . Now  $\Phi(G/G')$  is isomorphic to  $\upsilon_1(G)$  , which is cyclic; accordingly, either  $x^p\in G'$  or  $y^p\in G'$  . Thus, one of  $\alpha$  or  $\beta=1;$  a contradiction is reached. Hence,  $G'\cap\upsilon_1(G)\neq 1$  .

In view of Proposition 1.6,  $R_0' \leq \upsilon_1(G)$ . Then  $(G: \upsilon_1(G)) = p^3$ . It now follows that  $G/R_0'$  is nonmetacyclic and Redei.  $\Phi(R_0/R_0')$  is isomorphic to  $\upsilon_1(R_0)$  and has index  $p^2$  in  $R_0/R_0'$ . Since  $R_0/R_0'$  is abelian,  $\Omega_1(R_0/R_0')$  has order  $p^2$ . But, since  $G/R_0'$  is nonmetacyclic,  $\Omega_1(G/R_0')$  has order  $p^3$ . Thus, there exists an element  $\overline{d} \in G/R_0'$ 

such that  $\overline{d}^p=1$  and  $G/R_0'=(R_0/R_0')<\overline{d}^p>$ . Then, there exists a  $d \in G \wedge R_0$  such that  $dR_0'=\overline{d}$ . Hence,  $d^{p^2}=1$  and  $d^p \in R_0'=< c>$ . It may be assumed that  $d^p=c$  ( If not, then  $d^p=c^\alpha$  where  $\alpha \not\equiv 0 \pmod p$ ). There exists a  $\gamma$  such that  $\gamma \alpha \equiv 1 \pmod p$  and  $(d^\gamma)^p=c$ . Then d can be replaced by  $d^\gamma$ ). Consequently,

$$G = R_0 < d | d^p = 1 > \text{ where } d^p = c$$
 .

Moreover, there is a maximal subgroup M such that d  $\epsilon$  M . If M is nonabelian, then d<sup>p</sup> = R'<sub>0</sub>  $\cap$   $\upsilon_1(M)$  = M'  $\cap$   $\upsilon_1(M)$  = 1 , which is a contradiction. Thus, M is abelian, that is, M = M<sub>0</sub> . Since b  $\epsilon$   $\Omega_1(G) \leq M_0$  , then

$$[b,d] = 1$$
.

If  $[a,d] \in R_0'$ , then it follows from  $G/R_0' = (R_0 < d >)/R_0'$  that  $G/R_0'$  is abelian, which is a contradiction. Thus,

$$[a,d] = b^r d^{Sp}$$
,  $r \not\equiv 0 \pmod{p}$ .

Let  $d_1=d^t$  where t is such that  $\text{tr} \in 1 \pmod{p}$ .  $[a,d] \in M_0$ , so that [[a,d],d]=1. Then by Proposition 1.5,  $[a,d_1]=bd_1^{\text{Sp}}$ . Let  $b_1=bd_1^{\text{Sp}}$ . Then  $b_1 \not\in Z(G)=\Phi(R_0)$  and  $b_1 \in \Omega_1(G) \leq \Phi(G)$ . Also,  $[a,b_1]=d_1^{\text{rp}}$ . Hence,

$$G = R_0 < d_1 | d_1^p = 1 > \text{ where } < a, b_1 | a^p^m = b_1^p = c^p = 1, c = [a, b_1], [a, c] = [b_1, c] = 1 >$$

$$[a, d_1] = b_1, [b_1, d_1] = 1 \text{ and } c = d_1^{p} \text{ for } r \not\equiv 0 \pmod{p} .$$

This is the group of type (8) in Theorem 3.1.

Lemma 3.5: Given hypotheses (3.1). If (G:  $\Phi(G)) = p^3$  , then  $|\Omega_1(G)| = p^3 \; .$ 

Proof: By Lemma 3.2,  $p^3 \leq |\Omega_1(G)| \leq p^4$ . Suppose that  $|\Omega_1(G)| = p^4$ . By Lemma 3.1, G is regular. Then,  $(G: \upsilon_1(G)) = p^4$  and  $(\Phi(G): \upsilon_1(G)) = p$ . Since  $(M: \upsilon_1(G)) = p^3$  for each nonabelian maximal subgroup M, then  $\upsilon_1(G) = \upsilon_1(M)$  and M is nonmetacyclic. Now,  $M' \cap \upsilon_1(G) = M' \cap \upsilon_1(M) = 1$  so that  $\overline{G} = G/\upsilon_1(G)$ , which has order  $p^4$ , has precisely one abelian maximal subgroup  $M_0/\upsilon_1(G) = \overline{M}_0$ .  $\overline{G}'$  has order p and  $\overline{G}' = \overline{M}' = Z(\overline{M})$  for each nonabelian maximal subgroup  $\overline{M}$  of  $\overline{G}$ .

Let  $x \in G \sim M_0$ ; then  $\Phi(\overline{G}) < \overline{x} >$  has order  $p^2$ , is elementary abelian, and is normal in  $\overline{G}$ . Denote  $\Phi(\overline{G}) < \overline{x} >$  by  $\overline{A}$ . By Proposition 1.4(a),  $(\overline{G} : C_{\overline{G}}(\overline{A})) \le p$ ; also,  $\overline{A} \le Z(C_{\overline{G}}(\overline{A}))$ . If  $\overline{G} = C_{\overline{G}}(\overline{A})$ , then  $\overline{x} \in Z(\overline{G})$ . It follows from  $\overline{G} = \overline{M}_0 < \overline{x} >$  that  $\overline{G}$  is abelian, which is a contradiction. Thus  $C_{\overline{G}}(\overline{A})$  is maximal in  $\overline{G}$ . Since  $(C_{\overline{G}}(\overline{A}) : \overline{A}) = p$ ,  $C_{\overline{G}}(\overline{A})$  is abelian. Consequently,  $C_{\overline{G}}(\overline{A}) = \overline{M}_0$ , which implies that  $x \in M_0$ ; a contradiction is reached. Hence,  $|\Omega_1(G)| = p^3$ .

Lemma 3.6: Given hypotheses (3.1). If  $(G: \Phi(G)) = p^3$  and if  $G' = R_0'$ , then  $\Phi(G)\Omega_1(G)$  is a maximal subgroup of G. Furthermore, G is the group of type (9).

Proof:  $\Omega_1(G)$  is clearly a proper subgroup of G . Thus, three cases must be considered.

<u>Case 1</u>: Suppose that  $\Omega_1(G) \leq \Phi(G)$ . Then each maximal subgroup

of G is nonmetacyclic. Let  $x \in G \wedge M_0$ . There exists a maximal subgroup R to which x belongs. R is nonabelian, which implies that  $R' \cap \upsilon_1(R) = 1$ . Let  $\overline{G} = G/\upsilon_1(R_0)$ , which has order  $p^4$ .  $\varphi(\overline{G}) < \overline{x} > 1$  has order  $p^2$ , is elementary abelian, and is normal in  $\overline{G}$ . Let  $\overline{A} = \varphi(\overline{G}) < \overline{x} > 1$  If  $\overline{A} \leq Z(\overline{G})$ , then  $\overline{A} \leq Z(\overline{R})$  since  $\overline{A} < \overline{R}$ ; consequently,  $\overline{R}$  is abelian, which is a contradiction to  $R' \cap \upsilon_1(R) = 1$ . Hence,  $\overline{A} \not\leq Z(\overline{G})$ . By Proposition 1.4(a),  $C_{\overline{G}}(\overline{A})$  has index p in  $\overline{G}$ . But  $\overline{A} \leq Z(C_{\overline{G}}(\overline{A}))$ , whence  $C_{\overline{G}}(\overline{A})$  is abelian. Thus,  $\overline{x} \in \overline{M}_0$  from which it follows that  $x \in M_0$ ; a contradiction is reached. Thus  $\Omega_1(G) \not\leq \varphi(G)$ .

<u>Case 2</u>: Suppose that  $\Phi(G)\Omega_1(G)$  has index  $p^2$  in G. Then  $\Phi(G)\Omega_1(G)$  is abelian and nonmetacyclic. Denote  $\Phi(G)\Omega_1(G)$  by A. There exists a maximal subgroup  $A^*$  which contains A.  $A^*$  is nonmetacyclic since A is nonmetacyclic.

The regularity of G implies that  $(G: \upsilon_1(G)) = p^3$ . Thus,  $G' \leq \upsilon_1(G)$ . If each element x, for which  $x^p \in G'$ , is contained in A, then  $G' \leq \upsilon_1(A) \leq \upsilon_1(A^*)$ . If  $A^*$  were nonabelian, then  $A^*' \cap \upsilon_1(A^*) = G' \cap \upsilon_1(A^*) \neq 1$ , which is a contradiction. Therefore,  $A^*$  is abelian. If, on the other hand, there exists an  $x \in G$  such that  $x \not\in A$  but such that  $x^p \in G'$ , then A < x > is maximal in <math>G, is nonmetacyclic, and  $G' \leq \upsilon_1(A < x >)$ . By the same argument used to show that  $A^*$  was abelian, it can be concluded that A < x > is abelian. In either situation, A is contained in the abelian maximal subgroup  $M_O$ .

Suppose that there exists a nonmetacyclic nonabelian maximal subgroup  $N_0$ .  $N_0' \cap \upsilon_1(N_0) = 1$ . Let  $\overline{G} = G/\upsilon_1(N_0)$ , which has order  $p^4$ .  $\overline{G}'$  has order p and  $\overline{G}$  has precisely one abelian maximal subgroup  $\overline{M}_0$ . Now,  $\upsilon_1(\overline{G}) = \overline{\upsilon_1(G)}$ , which has order p. From the regularity of

 $\overline{G}$ ,  $|\Omega_1(\overline{G})| = p^3$ . Then,  $\overline{N}_0 = \Omega_1(\overline{G})$ . Let  $x \in N_0 \sim \Phi(G)$ .  $x^p \in \upsilon_1(N_0)$ , but  $\overline{x} \not\in Z(\overline{N}_0) = Z(\overline{G}) = \Phi(\overline{G})$ . By Proposition 1.4(a),  $C_{\overline{G}}(\Phi(\overline{G}) < \overline{x} >)$  has index p. Thus, it follows that  $C_{\overline{G}}(\Phi(\overline{G}) < \overline{x} >)$  is abelian, that is,  $C_{\overline{G}}(\Phi(\overline{G}) < \overline{x} >) = \overline{M}_0$ . In view of the fact that x is arbitrary in  $N_0 \sim \Phi(G)$ ,  $\overline{M}_0 = C_{\overline{G}}(\overline{x})$  for each  $x \in N_0 \sim \Phi(G)$ . However, for  $x \in \Phi(G) < M_0$ ,  $\overline{M}_0 = C_{\overline{G}}(\overline{x})$ . Thus,  $\overline{M}_0 = C_{\overline{G}}(\overline{N}_0)$ , which implies that  $\overline{M}_0 \cap \overline{N}_0 = Z(\overline{N}_0)$ . Hence,  $\overline{N}_0$  is abelian, and a contradiction is reached. Then,  $M_0$  is the only normetacyclic maximal subgroup, and  $M_0$  is the only maximal subgroup containing A.

Let R be any metacyclic maximal subgroup.  $\Omega_1(R) = \Omega_1(\Phi(G))$ , so that R has the defining relations R =  $< a,b \mid a^{p^m} = b^{p^n} = 1$ ,  $[a,b] = a^{p^{m-1}}$ , m, n  $\geq 2 >$ . G =  $< a^{p^{m-1}} >$  and  $\Phi(G) = \Phi(R) = < a^p > \otimes < b^p >$ . There exists an  $x \in M_0 \sim R$  such that  $x^p = 1$ . Then, A =  $\Phi(G) < x >$ . Now A< b > is a maximal subgroup of G. Thus, A< b > =  $M_0$ . However, A< a > is a maximal subgroup of G; so, A< a > =  $M_0$ . Hence, R =  $< a,b > = M_0$ , which is a contradiction. Consequently,  $(G: \Phi(G)\Omega_1(G)) \neq p^2$ .

 $\underline{\text{Case 3:}} \quad \phi(G)\Omega_1(G) \quad \text{is a maximal subgroup of } G \text{ . Denote } \\ \phi(G)\Omega_1(G) \quad \text{by M . Then, } \quad \Omega_1(G) = \Omega_1(M) \text{ . Since } |\Omega_1(G)| = p^3 \text{ , M is } \\ \text{not metacyclic. Now, } \quad |\phi(G)\cap\Omega_1(G)| = p^2 \text{ . Since } G \text{ is regular, then } \\ \text{by Theorem 1.6, each element of } \quad \Omega_1(G) \text{ has order } p \text{ . Therefore, } \\ \Omega_1(\phi(G)) = \phi(G)\cap\Omega_1(G) \text{ , which implies that } \phi(G) \text{ is cyclic. Let} \\ \text{y } \quad \phi(G) \text{ such that } \text{y}^p = 1 \text{ . G'< y > is both elementary abelian and } \\ \text{normal in } G \text{ , so that by Proposition 1.4(a), } \quad (G: C_G(G'< y >)) \leq p \text{ .} \\ \text{Since } \text{y } \notin Z(G) \text{ , } C_G(G'< y >) \text{ is a maximal subgroup of } G \text{ . But, } \\ \phi(G) < \text{y } > \leq Z(C_G(G'< y >)) \text{ and } \quad (C_G(G'< y >): \phi(G) < y >) = p \text{ , from} \\ \text{which it follows that } \quad C_G(G'< y >) \text{ is abelian, that is, } \quad C_G(G'< y >) = M_O \text{ .} \\ \end{aligned}$ 

Since y is arbitrary,  $\Omega_1(G) \leq M_0$ . Thus,  $\Phi(G)\Omega_1(G) = M_0$ .

Let R be a maximal subgroup of G , R  $\neq$  M<sub>0</sub> . R  $\cap$   $\Omega_1$ (G) has order p<sup>2</sup>. Therefore, R is metacyclic with  $\Phi$ (R) =  $\Phi$ (G) cyclic. Then, R = < a,b|ap<sup>m</sup> = b<sup>p</sup> = 1 , [a,b] = ap<sup>m-1</sup> > . There exists an element c  $\in$  M<sub>0</sub>  $\sim$  R such that c<sup>p</sup> = 1 and G = R< c > . Since b  $\in$   $\Omega_1$ (G)  $\leq$  M<sub>0</sub> , [b,c] = 1 . Moreover, G'  $\leq$  < a > , whereby < a >  $\triangleleft$  G . Now, < a > has index p<sup>2</sup> in G; then, < a >< c > is a maximal subgroup of G which is not abelian. Thus, [a,c] = a<sup>rp<sup>m-1</sup></sup> . Let t be such that tr = 1(mod p) and let c<sub>1</sub> = c<sup>t</sup> . G' < Z(G) so that by Proposition 1.5 [a,c<sub>1</sub>] = [a,c<sup>t</sup>] = [a,c)<sup>t</sup> = a<sup>trp<sup>m-1</sup></sup> . Hence,

$$G = R < c_1 | c_1^p = 1 > where R = < a,b | a^{p^m} = b^p = 1, [a,b] = a^{p^{m-1}} >$$
, 
$$[a,c_1] = a^{p^{m-1}} \text{ and } [b,c_1] = 1.$$

This is the group of type (9) in Theorem 3.1.

Lemma 3.7: Given hypotheses (3.1). If  $(G: \Phi(G)) = p^3$  and if  $R_0' < G'$ , then either  $\Omega_1(G) \le \Phi(G)$  or  $\Phi(G)\Omega_1(G)$  has index  $p^2$ . If  $\Omega_1(G) \le \Phi(G)$ , then G is the group of type (11). If  $\Omega_1(G) \le \Phi(G)$ , then G is the group of type (10).

Proof: Let A be a normal subgroup of G', minimal with the property that M' < A for each nonabelian maximal subgroup M . Each maximal subgroup of G/A is abelian, so that G/A is either Redei or abelian. If G/A is Redei, then  $(G/A: \Phi(G/A)) = p^2$ , which contradicts the hypothesis that  $(G: \Phi(G)) = p^3$ . Thus G/A is abelian. Hence,

 $G^{\dagger} = A$ . This implies that

(3.6) G has at least two maximal subgroups with distinct commutator subgroups.

Furthermore, G' is elementary abelian.

Since G is regular, then by Theorem 1.6, each element of  $\Omega_1(G) \text{ has order } p \text{ . Therefore, since } G' \leq \Omega_1(G) \text{ , } |\Phi(G) \cap \Omega_1(G)| \geq p^2$  and  $(\Omega_1(G) \colon \Phi(G) \cap \Omega_1(G)) \leq p \text{ . This implies that } (\Phi(G)\Omega_1(G) \colon \Phi(G)) \leq p$  and that  $\Phi(G)\Omega_1(G) \text{ is abelian.}$ 

Then,  $|\Omega_1(M)| = p^2$ , which implies that M is metacyclic. Thus,  $\upsilon_1(M) = \phi(G)$ . For each maximal subgroup  $M^*$  of M,  $G' \leq \phi(G) = \upsilon_1(M) < M^*$  so that  $M^* \triangleleft G$ . Hence,  $M^*\Omega_1(G)$  is a maximal nonmetacylic subgroup of G. M has p+1 maximal subgroups in all, each of which determines a nonmetacyclic maximal subgroup of G. G then has at least p+1 nonmetacyclic maximal subgroups, of which at most one could be abelian. Thus, let R be a nonabelian, nonmetacyclic maximal subgroup of G. Since  $\Omega_1(R) \not\leq \phi(G) = \phi(R)$ , then R is of the form

$$R = \langle a,b | ap^m = b^p = c^p = 1, c = [a,b], [a,c] = [b,c] = 1 \rangle$$
.

Suppose that the abelian maximal subgroup  $\,\rm M^{}_{0}\,$  is metacyclic. Then,  $\,\upsilon_{1}(\rm M^{}_{0})\,=\,\Phi(R)\,=\,<\,a^{\rm p}\,>\,\otimes\,<\,c\,>\,$  . By Theorem 1.6, there exist

elements x and  $a_1$  in  $M_0$  such that  $x^p = c$  and  $a_1^p = a^p$ . However,  $x \not\in R$  so that  $M_0 = \langle x \rangle \otimes \langle a_1 \rangle$ . Let  $R^* = \Phi(G) \langle x \rangle$ .  $R^* \neq G$  and  $(G:R^*) = p^2$ . Also,  $b \not\in R^*$ , since  $R^*$  is a subgroup of the metacyclic group  $M_0$ . Thus,  $R^* < b \rangle$  is a maximal subgroup of G, is nonmetacyclic, and is nonabelian. In view of the fact that  $b \not\in \Phi(G) = \Phi(R^* < b \rangle)$ , then  $|\Omega_1(\Phi(R^* < b \rangle))| = p^2$ . Since  $(R^* < b \rangle)' \cap \upsilon_1(R^* < b \rangle) = 1$ , it follows that  $\upsilon_1(R^* < b \rangle)$  is cyclic. But  $x^p \in \upsilon_1(R^* < b \rangle)$  and  $a^p \in \upsilon_1(R^* < b \rangle)$  so that  $x^p \in \Omega_1(\upsilon_1(R^* < b \rangle)) = \langle a^{p^{m-1}} \rangle$ . Thus,  $c \in \langle a^{p^{m-1}} \rangle$ , which is a contradiction unless m = 1. Consequently,  $|R| = p^3$ ,  $|G| = p^4$  and  $|G'| \leq |\Phi(G)| = p$ ; a contradiction is reached. Therefore, the abelian maximal subgroup  $M_0$  of G is nonmetacyclic.

Consider  $\overline{G}=G/\upsilon_1(R)$ , which is nonabelian of order  $p^4$ .  $\overline{R}$  is a nonabelian maximal subgroup of  $\overline{G}$ . Since  $|\upsilon_1(\overline{G})|=p$ , then by the regularity of  $\overline{G}$ , it follows that  $\Omega_1(\overline{G})=\Omega_1(\overline{R})=\overline{R}$ . Let  $R^*$  be a nonmetacyclic, nonabelian maximal subgroup different from R. If  $R^{*'}\leq \upsilon_1(R)$ , then  $\overline{R}^*\cap \overline{M}_0=Z(\overline{G})$ . Then,  $\overline{A}_0=Z(\overline{G})$ . But,  $\overline{A}_0\leq \overline{R}$ , so that  $\overline{R}$  is abelian, which is a contradiction. Hence,  $R^{*'}\not\leq \upsilon_1(R)$  and  $\overline{R}^*$  is nonabelian. Moreover,  $\Omega_1(\overline{R}^*)=\overline{R}^*\cap\Omega_1(\overline{G})$ . Thus,  $|\Omega_1(\overline{R}^*)|=p^2$ , from which it follows that  $\overline{R}^*$  is metacyclic. Therefore, there exists an element x in  $R^*$  such that  $(\overline{x}^p)=\overline{R}^{*'}=Z(\overline{R}^*)$ . Since  $\overline{x}\not\in Z(\overline{R}^*)$ ,  $\overline{x}\not\in Z(\overline{G})$ . By Proposition 1.4(a),  $C_{\overline{G}}((\overline{x}>))$  is maximal in  $\overline{G}$ . But  $(\overline{x}>\leq Z(C_{\overline{G}}((\overline{x}>))))$ ; thus,  $C_{\overline{G}}((\overline{x}>))$  is abelian. Since  $\overline{R}^*=(\overline{x}><\overline{D}>$ ,  $\overline{D}\not\in C_{\overline{G}}((\overline{x}>))$ . However,  $\overline{D}\in \overline{\Omega_1(\overline{G})}$ , which implies that  $C_{\overline{G}}((\overline{x}>))$  is not the image of any nonmetacyclic maximal subgroup of G. There exists, then, a metacyclic maximal subgroup M such that  $\overline{M}=C_{\overline{G}}((\overline{x}>))$ . Thus,  $M'=(\overline{D})$ 

From  $G = M_0M$ , it follows that Z(G/M') has index  $p^2$  and that  $Z(G/M') \not R_0/M'$ . There exists a  $y \in M$  such that  $y \not \in R$  and  $y^p = c \in R'$ . G' < y > is abelian of type (2,1) but  $y \not \in Z(G)$ . By Proposition 1.4(b),  $C_G(G' < y >)$  is a maximal subgroup of G.  $\Phi(G) < y > \leq Z(C_G(G' < y >))$ , which implies that  $C_G(G' < y >)$  is abelian, that is,  $C_G(G' < y >) = M_0$ . Thus,  $y \in M_0 \cap M$ . If  $\alpha$  is the natural homomorphism  $\alpha \colon G \to G/M'$ , then  $y^\alpha \in Z(G/M')$ . Furthermore,  $[a^\alpha, y^\alpha] = 1$ ,  $[b^\alpha, y^\alpha] = 1$ , and  $[c^\alpha, y^\alpha] = 1$ . Since both b and y belong in  $M_0$ , [b, y] = 1.  $[a, y] \in M' = \langle a \rangle^{m-1} > so$  that  $[a, y] = a^{m-1}$ ,  $r \not = 0 \pmod p$  (otherwise  $G' = R_0'$ ). Let s be such that  $rs \equiv 1 \pmod p$  and let  $y_1 = y^s$ . Then  $y_1^p = y^{sp} = c^s$ . Let  $c_1 = c^s$  and let  $b_1 = b^s$ . Since  $[a, b] = c \in G' \leq Z(G)$  and  $[a, y_1] \in G' \leq Z(G)$ , then by Proposition 1.5  $[a, b_1] = c_1$  and  $[a, y_1] = a^{m-1}$ . Thus,

$$G = S < b_1 | b_1^p = 1 > \text{where } S = < a, y_1 | a^{p^m} = y_1^{p^2} = 1, [a, y_1] = a^{p^{m-1}} > ,$$

$$[a, b_1] = y_1^p \text{ and } [b_1, y_1] = 1.$$

This is the group of type (10) in Theorem 3.1.

Case 2:  $\Omega_1(G) \leq \Phi(G)$ . Then, each maximal subgroup of G is nonmetacyclic. From the regularity of G,  $(G: \upsilon_1(G)) = p^3$  and  $\Phi(G) = \upsilon_1(G)$ . Let  $R_0$  and  $R_1$  be two nonabelian maximal subgroups, with  $R_0' \neq R_1'$ . Two such subgroups exist by (3.6). Consider  $\overline{G} = G/R_0'$ . This factor group has two abelian maximal subgroups  $\overline{R}_0$  and  $\overline{M}_0$ , so that  $Z(\overline{G})$  has index  $p^2$  in  $\overline{G}$ . It follows that  $\overline{G}$  has at least p+1 abelian maximal subgroups. Let  $A_1$  be a maximal subgroup of  $R_1$ .

 $R_0^{'} \leq \Phi(G) = \Phi(R_1) \leq A_1$  , and  $Z(\overline{G}) \ \overline{A}_1$  is abelian maximal in  $\overline{G}$ . Thus, each maximal subgroup of  $R_1$  determines an abelian maximal subgroup of  $\overline{G}$ . Conversely, if  $\overline{M}$  is an abelian maximal subgroup of  $\overline{G}$ , then  $Z(\overline{G}) \leq \overline{M}$  (otherwise,  $\overline{G}$  would be abelian).  $\overline{M} \cap \overline{R}_1$  is maximal in  $\overline{R}_1$ , which is nonabelian and Redei. Thus,  $(\overline{M} \cap \overline{R})Z(\overline{G}) = Z(\overline{G})\overline{R}_1 \cap \overline{M} = \overline{G} \cap \overline{M} = \overline{M}$ . Hence, there is a one-to-one correspondence between the maximal abelian subgroups of  $\overline{G}$  and the maximal subgroups of  $R_1$ , that is,  $\overline{G}$  has exactly p+1 abelian maximal subgroups. Therefore, G has exactly p nonabelian maximal subgroups with the commutator subgroup  $R_0^{'}$ . In view of the fact that  $R_0$  is arbitrary, the nonabelian subgroups of G are separated into classes where the subgroups in each class all have the same commutator subgroup. Moreover, there are p subgroups in each class. Since there are  $p^2+p+1$  maximal subgroups [15, p. 311] of which  $p^2+p$  are nonabelian, there are p+1 distinct classes, that is, p+1 distinct commutator subgroups.

Let  $N_0$  be a nonabelian maximal subgroup.  $N_0 \cap M_0$  has index  $p^2$  in G and is normal in G. There exists a  $y \in N_0 \sim M_0$  such that  $y^p \neq 1$ .  $N_0 = (M_0 \cap N_0) < y >$  and  $G = M_0 < y >$ . Also, there exists an  $x \in M_0 \sim N_0$  such that  $x^p \neq 1$ ,  $M_0 = (M_0 \cap N_0) < x >$  and  $G = N_0 < x >$ . If [x,y] = 1, then  $x \in Z(G) = \Phi(G) \leq N_0$ , which is a contradiction. Thus,  $[x,y] \neq 1$ . Now,  $N_0' < [x,y] > \leq G'$  and  $|N_0' < [x,y] > | \leq p^2$ . Since  $G' \leq Z(G)$ , then  $N_0' < [x,y] > \triangleleft G$ . Let  $\alpha$  be the natural mapping  $\alpha \colon G \to G/N_0' < [x,y] >$ .  $G^\alpha = G/N_0' < [x,y] >$ .  $G^\alpha = M_0^\alpha < y^\alpha > = ((M_0 \cap N_0)^\alpha < x^\alpha >) < y^\alpha >$ . But,  $(M_0 \cap N_0)^\alpha < y^\alpha > = N_0^\alpha$ , which is abelian. This implies that  $(x,y)^\alpha > (x,y)^\alpha > (x,y)^\alpha$ 

follows that  $G^{\alpha} = M_0^{\alpha} < y^{\alpha} > is abelian, that is, <math>G^{'} \leq N_0^{'} < [x,y] > .$  In addition, since  $|G^{'}| \geq p^2$ , then  $[x,y] \not\in N_0^{'}$  and  $G^{'} = N_0^{'}[x,y]$ .

Since G is regular, then by Theorem 1.6, each element of  $\upsilon_1(G)$  is the  $p^{th}$  power of an element of G . Since  $\Phi(N_0) = \upsilon_1(G)$ , there exists a z  $\epsilon$  G such that  $< z^P > = N_0'$ . However,  $N_0' \cap \upsilon_1(N_0) = 1$  whereby z  $\not\in N_0$ . Now, G' < z > is normal in G , is abelian of type (2,1) but is not contained in  $Z(G) = \Phi(G)$ . Thus,  $C_G(G' < z >)$  is a maximal subgroup of G by Proposition 1.4(b). Furthermore, Z(G) < z > has index p in  $C_G(G' < z >)$  and  $Z(G) < z > \leq Z(C_G(G' < z >))$ . These imply that  $C_G(G' < z >)$  is abelian. Thus,  $M_0 = C_G(G' < z >)$ ; and accordingly,  $N_0' \leq \upsilon_1(M_0)$ . From the fact that  $N_0$  is an arbitrary nonabelian subgroup of G , it follows that  $G' \leq \upsilon_1(M_0)$ .

For each subgroup  $Z_1$  (for  $i=1,2,\cdots,p+1$ ) of G' of order p, there exists, by the argument used in the paragraph above, an element  $z_i \in \mathbb{M}_0 \rightsquigarrow \Phi(G)$  such that  $z_1^{p^2}=1$ , and such that  $Z_1=\langle z_1^p \rangle$ . Let  $\mathbb{M}_0^*=\langle z_1|i=1,2,\cdots,p+1 \rangle$ . Then,  $\mathbb{M}_0^*$  is abelian but is not contained in  $\Phi(G)$ . The exponent of  $\mathbb{M}_0^*$  is  $p^2$ , and  $G'=\Omega_1(\mathbb{M}_0^*)=\upsilon_1(\mathbb{M}_0^*)$ . Therefore, from the regularity of G,  $(\mathbb{M}_0^*:\upsilon_1(\mathbb{M}_0^*))=p^2$ ,  $|\mathbb{M}_0^*|=p^4$  and  $\mathbb{M}_0^*$  has exactly p+1 maximal subgroups. Each maximal subgroup of  $\mathbb{M}_0^*$  corresponds to a distinct commutator subgroup of a nonabelian maximal subgroup of G; that is,  $G'< z_1 \rangle$ , for each  $i=1,2,\cdots,p+1$ , is a maximal subgroup of  $\mathbb{M}_0^*$  is properly contained in  $\mathbb{M}_0$ , then  $\Phi(G)\cap\mathbb{M}_0^*$  is a maximal subgroup of  $\mathbb{M}_0^*$ . But then,  $\Phi(G)\cap\mathbb{M}_0^*=G'< z_1 \rangle$ , where  $\{z_1^p\}=Z_1\leq G'$  for some  $\{1,2,\cdots,p+1\}$ . Consequently,  $\{z_1^p\}=\Phi(G)\}$ , which is a contradiction. Thus,  $\{0,0\}$ , and  $\{0,0\}$  is contained in no nonabelian maximal subgroup of  $\{0,0\}$ . Then, for a nonabelian

maximal subgroup  $N_0$ ,  $G = N_0 M_0^*$ .  $(M_0^*: N_0 \cap M_0^*) = p$ , so that  $N_0 \cap M_0^*$  is a maximal subgroup of  $M_0^*$ . This implies that  $N_0 \cap M_0^* = G' < z > for some <math>z \in M_0 \land \Phi(G)$  of order  $p^2$ . Hence,  $z \in N_0$ , and  $(z^p) = N_1'$  for some nonabelian maximal subgroup  $N_1$  different from  $N_0$ .

Suppose that  $N_0 = a_1b_1a_2^m = b_1^p = c^p = 1$ ,  $c = [a_1b]$ ,  $[a_1c] = [b_1c] = 1 > .$  Since  $\Omega_1(G) = \Omega_1(N_0) \le \Phi(N_0)$ , then both  $m, n \ge 2$ . Let  $z = a^rb^sc^t$ .  $z \not\in \Phi(G)$ , so that at least one of  $r, s \not\in 0 \pmod p$ ; say  $s \not\in 0 \pmod p$ .  $1 = z^{p^2} = (a^rb^s)^{p^2}$ . By Definition 1.4,  $(a^rb^s)^{p^2} = a^{rp^2}b^{sp^2}\Pi dp^2$  where  $d_1 \in (a_1b)^p$ . Thus,  $(a^rb^s)^{p^2} = a^{rp^2}b^{sp^2} = 1$ , in from which it follows that  $a^{rp^2} = b^{-sp^2}$ . Then,  $r \equiv 0 \pmod p^{m-2}$  and  $s \equiv 0 \pmod p^{n-2}$ . But then,  $s \equiv 0 \pmod p$ , which is a contradiction unless n = 2. Therefore,  $z = a^{r^*p^{m-2}}b^sc^t$  and  $b = (a^{-r^*p^{m-2}}zc^{-t})^{-s}$  for some  $r^*$ . Let  $c_1 = [a_1z] = c^s$ . It follows that

$$N_0 = \langle a, z | a^{p^m} = z^{p^2} = c_1^p = 1, c_1 = [a, z], [a, c_1] = [z, c_1] = 1 \rangle$$
.

By the argument used earlier in the proof, there exists an element  $\omega$   $\epsilon$  M  $_0$   $\sim$  N  $_0$  for which  $\omega^p{}^2$  = 1 and  $\omega^p$  =  $c_1$  . Then,

$$G = N_0 < \omega |_{\omega}^{p^2} = 1 > .$$

Since both z and  $\omega$  are in M $_{\cap}$ ,

$$[z,\omega] = 1$$
.

Now,  $[\omega,a] \in G' = \langle z^p,c_1 \rangle$  but  $[\omega,a] \not\in \langle c_1 \rangle$  (otherwise  $\langle \omega \rangle \triangleleft G$  and  $\langle \omega \rangle \langle a \rangle$  is a nonabelian subgroup of index  $p^2$  in G).

Let

$$[\omega,a] = c_1^u z^{vp}$$
 where  $v \neq 0$ .

If  $u \not\equiv 0 \pmod p$  , then there exists an x such that  $xu \equiv 1 \pmod p$  . Let

$$a_1 = a^X$$
 and  $z_1 = z^U$ .

Then  $[a_1,z_1] \in \mathbb{N}_0^i \leq Z(G)$ , so that by Proposition 1.5,  $[a_1,z_1] = c_1$ . Then

$$N_0 = \langle a, z_1 | a_1^{p^m} = z_1^{p^2} = c_1^{p} = 1, c_1 = [a_1, z_1], [a_1, c_1] = [z_1, c_1] = 1 \rangle$$

Also,  $[\omega, a_1] \in G' \leq Z(G)$ , which implies that  $[\omega, a_1] = c_1 z_1^{X^2 vp}$ . Hence,

$$G = N_0 < \omega | \omega^{p^2} = 1 > \text{ where } \omega^p = c_1 \text{ and }$$

(3.8)

$$[\omega, a_1] = c_1 z^{kp}$$
 for  $k \neq 0 \pmod{p}$ .

If u = 0, then

$$G = N_0 < \omega | \omega^{p^2} = 1 > \text{ where } N_0 = < a, z | a^{p^m} = z^{p^2} = c^p = 1, c = [a, z] > ,$$
(3.9)

$$\omega^p = c$$
 and  $[\omega, a] = z^{p}$  for  $r \not\equiv 0 \pmod{p}$ .

The groups in (3.8) and (3.9) are the groups of type (10) in Theorem 3.1.

Proof of THEOREM 3.1: Let  $M_0$  be the abelian maximal subgroup and let  $R_0$  be the Redei maximal subgroup. By Proposition 1.1,

 $Z(R_0) \le \Phi(G) \le M_0$ . It follows from  $G = R_0 M_0$  that  $Z(R_0) \le Z(G)$ . But  $Z(G) \le Z(R_0)$ , since  $G \in \mathcal{J}$ . Thus,  $Z(G) = Z(R_0)$  and  $(G: Z(G)) = p^3$ . Furthermore, by Proposition 3.3, each Redei subgroup R is a maximal subgroup and Z(R) = Z(G).

If G is metacyclic, then by Theorem 2.4, G  $\epsilon$   $R^*$ ; so, each maximal subgroup is a Redei group. This is a contradiction to the fact that  $M_{\Omega}$  is abelian. Hence, G is not a metacyclic group.

Two possible situations arise. Either each maximal subgroup of G is metacyclic or there is at least one nonmetacyclic maximal subgroup. The case where there is a nonmetacyclic maximal subgroup has been examined in Lemmas 3.1 through 3.7, and the groups have been characterized.

For the case where each maximal subgroup is metacyclic, by Theorem 1.10, it follows that G is a 3-group of class 3 and order  $3^4$ . By Theorem 1.9 (G:  $\upsilon_1(G)$ ) =  $p^3$ . Thus,  $\upsilon_1(G)$  = Z(G) = Z(R) =  $\upsilon_1(R)$  for each Redei subgroup. Also, since the class of G is 3,  $|G^{\dagger}|$  =  $3^2$ .

If there exists an element  $x \not\in G \circ G'$  such that  $x^p = 1$ , then there exists a maximal subgroup M such that  $x \not\in M$ .  $|\Omega_1(M)| = 3^2$ . Thus,  $|\Omega_1(M)| = 3^3$ . But  $\Omega_1(\Omega_1(M) < x >) = \Omega_1(M) < x >$ , so that  $\Omega_1(M) < x >$  is not metacyclic, which is a contradiction. Hence,  $G' = \Omega_1(G)$ . Then,  $U_1(G) = \langle a^3 \rangle$  and  $G' = \langle a^3, b \rangle$ . There is an element  $C \in M_0 \circ R_0$  such that  $C^{3^2} = 1$ . Then,

 $G = R_0 < c | c^{3^2} = 1 >$ , [b,c] = 1 and  $c^3 = a^{3r}$  for  $r \not\equiv 0 \pmod{3}$ . Now,  $[a,c] = b^s a^{3t}$  where  $s \not\equiv 0 \pmod{3}$ . There exists  $s^*$  such that  $ss^* \equiv 1 \pmod{3}$ . Let

$$c_1 = c^{s^*}$$
 and  $b_1 = bx^{3st}$ .

Then, by Proposition 1.5,  $[a,c_1] = b_1$ . Also,  $[a,b_1] = a^3$ , and  $[b_1,c_1] = 1$ . Hence,

$$G = R_0 < c_1 | c_1^{3^2} = 1 > \text{where } R_0 = < a, b_1 | a^{3^2} = b^3 = 1, [a, b_1] = a^3 > ,$$

$$[c_1, b_1] = 1, [a, c_1] = b_1 \text{ and } c_1^3 = a^{3u} \text{ where } u \neq 0 (\text{mod } 3) .$$

This is the group of type (1) in Theorem 3.1.

Each of the eleven types has been constructed from a specific case which relates the characteristic subgroup  $\,\Omega_1(G)\,$  to the characteristic subgroup  $\,\Phi(G)\,$ . It is clear, then, that these groups are pairwise nonisomorphic.

It should be noted that the group of type (2) in Theorem 3.1 is the group  $C_3$   $C_3$  of Example 1.1.

## CHAPTER IV

## ALL SUBGROUPS OF A FIXED INDEX ARE REDEI

In this chapter the groups in  $\mathcal{J}^* \sim \mathcal{R}$  with a Redei maximal subgroup are characterized. It follows as a consequence of this characterization that class  $\mathcal{R}$  is "almost equal" to class  $\mathcal{J}$ . Furthermore, for p > 2 a second characterization is found for the nonabelian metacyclic groups.

THEOREM 4.1: If G  $\epsilon$  3 \*  $\circ$  R and if G has a Redei maximal subgroup, then

- (a) G has a normetacyclic maximal subgroup,
- (b)  $Z(G) = v_1(G)$ ,
- (c) G' is elementary abelian of order  $p^3$ ,
- (d)  $(G: \Phi(G)) = p^2$ ,
- (e) no two Redei subgroups have the same commutator subgroup.

Proof: Since G has a Redei maximal subgroup, then by Definition 1.3, each maximal subgroup of G is Redei. Since G  $\not\in R$ , there exists a Redei subgroup  $R_0$  such that  $R_0 < C_G(Z(R_0))$ . Then,  $G = C_G(Z(R_0))$  and  $Z(R_0) \le Z(G)$ . But Proposition 1.3 implies that  $Z(G) \le Z(R_0)$ . Thus, Z(G) has index  $p^3$  in G. In addition, it follows that Z(G) = Z(R) for each Redei subgroup R of G.

If all the maximal subgroups of G are metacyclic, then by Theorem 1.10 either G is itself metacyclic or  $|G|=3^4$ . However, if G is metacyclic, then by Theorem 2.4, G  $\epsilon$   $R^*$ , which is a contradiction

to  $G \not\in \mathcal{R}$  . If, on the other hand,  $|G| = 3^4$ , then G has an abelian maximal subgroup by Proposition 1.4(c), which is a contradiction to  $G \in \mathcal{J}^*$ . Hence, G has at least one nonmetacyclic maximal subgroup, say  $N_0$ . Since  $N_0$  is Redei, then  $(N_0: v_1(N_0)) = p^3$  and  $N_0 \cap v_1(N_0) = 1$ .

To show that  $Z(G) = \upsilon_1(G)$ , first suppose that  $\upsilon_1(G) = \upsilon_1(N_0)$ . Since each maximal subgroup M is a Redei group and since  $(M: \upsilon_1(M)) \leq p^3$ , it then follows that  $\upsilon_1(G) = \upsilon_1(M)$ . Thus,  $(M: \upsilon_1(M)) = p^3$  for each maximal subgroup M . Also,  $N_0' \cap \upsilon_1(G) = 1$ , which implies that  $G/\upsilon_1(G)$  is nonabelian. By Proposition 1.4(c),  $G/\upsilon_1(G)$  has an abelian maximal subgroup  $M_0/\upsilon_1(G)$ . Then,  $M_0' \leq \upsilon_1(G) = \upsilon_1(M_0)$ ; consequently,  $\Phi(M_0) = \upsilon_1(M_0)$ , and  $(M_0: \Phi(M_0)) = p^3$ . This is a contradiction to the fact that  $M_0$  is a Redei subgroup. Therefore,  $\upsilon_1(N_0) < \upsilon_1(G)$ . Moreover, for each  $x \in G$ , there exists a maximal subgroup M such that  $x \in M$ . Then  $x^p \in \upsilon_1(M) \leq \Phi(M) = Z(M) = Z(G)$ . Thus,  $\upsilon_1(G) \leq Z(G)$ . Since  $(Z(G): \upsilon_1(N_0)) = p$ , then  $\upsilon_1(G) = Z(G)$ .

Now  $\upsilon_1(N_0)$ , as a characteristic subgroup of  $N_0$ , is normal in G.  $G/\upsilon_1(N_0)$  is nonabelian of order  $p^4$ , but by Proposition 1.4(c),  $G/\upsilon_1(N_0)$  has an abelian maximal subgroup  $M_0/(\upsilon_1(N_0))$ . Since  $N_0'\cap \upsilon_1(N_0)=1$ , it follows that  $N_0'M_0'$  has order  $p^2$ . Denote  $N_0'M_0'$  by A. A is normal in G since both  $N_0'$  and  $M_0'$  are characteristic subgroups of  $N_0$  and  $M_0$ , respectively. G/A has at least two abelian maximal subgroups  $N_0/A$  and  $M_0/A$ . Also, Z(G/A) has index  $p^2$  in G/A.

If G/A is abelian, then  $G^{'}=A$ . But  $A\leq Z(G)$  which implies that the class of G is 2. By Theorem 1.7(a), G is regular. Then, by Theorem 1.6(b), each element in  $\upsilon_1(G)$  is the  $p^{th}$  power of an element in G. In particular, since  $N_0^{'}\leq \upsilon_1(G) \, \sim \, \upsilon_1(N_0)$ , there exists an  $x\in G \, \sim \, N_0$  such that  $N_0^{'}=< x^p>$ . In view of the fact that  $G^{'}\leq Z(G)$ ,

 $G' < x > \triangleleft G$ ; then G' < x > is abelian of type (2,1). By Proposition 1.4(b),  $C_G(G' < x >)$  has index at most p in G. Now, since  $Z(G) = Z(N_O)$ , then  $x \not\in Z(G)$ ; therefore,  $C_G(G' < x >)$  is a maximal subgroup of G. However,  $Z(G) < x > \leq C_G(G' < x >)$  with index p. It follows that  $C_G(G' < x >)$  is an abelian maximal subgroup, which is a contradiction. Thus,

# (4.1) G/A is nonabelian and $|G'| \ge p^3$ .

Suppose that  $(G: \Phi(G)) = p^3$ , that is,  $\Phi(G) = \upsilon_1(G) = Z(G)$ . Then,  $G' \leq Z(G)$  and the class of G is 2. By Theorem 1.7(a), G is a regular group.

Let  $\overline{A}$  be a subgroup of G' minimal with the property that it contains the commutator subgroup of each of the maximal subgroups.  $A \leq \overline{A} \leq G'$ .  $\overline{A} \leq Z(G)$  so that  $\overline{A} \triangleleft G$ . Each maximal subgroup of  $G/\overline{A}$  is abelian, which implies that  $G/\overline{A}$  is either an abelian group or a Redei group. If  $G/\overline{A}$  is a Redei group, then  $\Phi(G/\overline{A})$  has index  $p^2$  in  $G/\overline{A}$ . But,  $(G: \Phi(G)) = (G/\overline{A}: \Phi(G)/\overline{A}) = (G/\overline{A}: \Phi(G/\overline{A}))$ . This is a contradiction to  $(G: \Phi(G)) = p^3$ . Thus,  $G/\overline{A}$  is abelian and  $\overline{A} = G'$ .

Since G/A is nonabelian, A <  $\overline{A}$  . Therefore, there exists a maximal subgroup R of G such that R'  $\neq$  A . Since G' < Z(G) , then AR' is elementary abelian of order p<sup>3</sup> . Also, AR'  $\leq$  N<sub>0</sub> , which implies that AR'  $\leq$   $\Omega_1(N_0)$  . Let S be a maximal subgroup of G different from R, N<sub>0</sub>, and M<sub>0</sub> . AR'S'  $\leq$   $\overline{A}$  < N<sub>0</sub> and AR'S' is elementary abelian so that AR'S'  $\leq$   $\Omega_1(N_0)$  . However,  $|\Omega_1(N_0)| = p^3$  since N<sub>0</sub> is normetacyclic. Therefore, AR'S' = AR' . Since S is any maximal subgroup, AR' is a minimal subgroup of G' containing the commutator subgroup of each maximal subgroup. Hence, AR' =  $\overline{A}$  = G' . Thus,  $|G'| = p^3$  and G'

is elementary abelian.

 $G' < M \quad \text{for each maximal subgroup so that} \quad G' \leq \Omega_1(M) \leq \Omega_1(G) \; .$  However, since G is a regular group and since  $(G: \upsilon_1(G)) = p^3$ , then  $|\Omega_1(G)| = p^3$ . Thus,  $G' = \Omega_1(M) = \Omega_1(G)$ . It follows, therefore, that  $(M: \upsilon_1(M)) = p^3$  and that M is nonmetacyclic. Each maximal subgroup, then, is a nonmetacyclic subgroup.

The fact that  $G/\Phi(G)$  is elementary abelian of order  $p^3$  implies that  $G/\Phi(G)$  has  $p^2+p+1$  maximal subgroups [15, p. 311]. Thus, G has  $p^2+p+1$  maximal subgroups. If M'=N' for two distinct maximal subgroups M and M, then by Theorem 1.6(b), there exists an  $x \in G$  such that  $M'=< x^p>$ . Both M and M are nonmetacyclic, which implies that  $x \not\in M$  and  $x \not\in N$ ; in particular,  $x \not\in Z(G) \le M \cap M$ . However,  $(x \times A) \cap M$  and  $(x \times A) \cap M$ . It follows that  $(x \times A) \cap M$  are Proposition 1.4(b),  $(G) \cap M \cap M$  is a maximal subgroup of  $(G) \cap M \cap M$  and  $(G) \cap M \cap M$  is a maximal subgroup of  $(G) \cap M \cap M$  and  $(G) \cap M \cap M$  is an abelian maximal subgroup of  $(G) \cap M \cap M$  is a contradiction to  $(G) \cap M \cap M$  is an abelian maximal subgroup of  $(G) \cap M \cap M$  is a contradiction to  $(G) \cap M \cap M$  is an abelian maximal subgroup of (G)

Now,  $G/N_0'$  has precisely one abelian maximal subgroup, and  $p^2+p$  nonabelian maximal subgroups. Let  $M_0$  be a maximal subgroup different from  $N_0$ .  $M_0' \leq \upsilon_1(G)$ , so there exists, by Theorem 1.6, an element  $x \in G$  such that  $< x^p > = M_0'$ . In addition,  $M_0$  is nonmetacyclic which implies that  $x \not\in M_0$ . Denote  $G/N_0'$  by  $\overline{G}$ .  $\overline{x}$  has order  $p^2$  since  $M_0' \neq N_0'$ . Moreover,  $\overline{G}'$  has order  $p^2$  and  $\overline{G}' < \overline{x} >$  has order  $p^3$ . Since  $G' \leq Z(G)$ , it follows that  $\overline{G}' < \overline{x} >$  is abelian of type (2,1). By Proposition 1.4(b),  $C_{\overline{G}}(\overline{G}' < \overline{x} >)$  has index at most p. If  $\overline{G}' < \overline{x} > \leq Z(\overline{G})$ , then  $Z(\overline{G}) = \overline{Z(G)} < \overline{x} >$ , which has index  $p^2$  in  $\overline{G}$ .

This implies that  $Z(\overline{G})$  is contained in no maximal subgroup other than  $\overline{N}_0$ . Let  $\overline{y}$  be any element of  $\overline{G} \sim Z(\overline{G})$ . If  $\overline{y}^D = 1$ , then  $Z(\overline{G}) < \overline{y} > 1$  is maximal in  $\overline{G}$ ; so  $Z(\overline{G}) < \overline{y} > 1$ , then since  $\Phi(\overline{G}) \leq Z(\overline{G})$ ,  $Z(\overline{G}) < \overline{y} > 1$  is maximal in  $\overline{G}$ , and  $Z(\overline{G}) < \overline{y} > 1$ , then since  $\Phi(\overline{G}) \leq Z(\overline{G})$ , which is a contradiction. Hence,  $\overline{G}' < \overline{x} > \not \leq Z(\overline{G})$  from which it follows that  $C_{\overline{G}}(\overline{G}' < \overline{x} >)$  is maximal in  $\overline{G}$ . However,  $Z(\overline{G}) < \overline{x} > \leq Z(C_{\overline{G}}(\overline{G}' < \overline{x} >))$  and  $(C_{\overline{G}}(\overline{G}' < \overline{x} >) : Z(\overline{G}) < \overline{x} >) = p$ . Consequently,  $C_{\overline{G}}(\overline{G}' < \overline{x} >)$  is abelian. This implies that  $x \in N_0$ , that is,  $M_0' \leq \upsilon_1(N_0)$ . Since  $M_0$  is arbitrary,  $M' \leq \upsilon_1(N_0)$  for each of the  $p^2+p$  maximal subgroups different from  $N_0$ .

Since  $N_0' \cap \upsilon_1(N_0) = 1$ ,  $\Omega_1(\upsilon_1(N_0))$  is elementary abelian of order  $p^2$ . Then,  $\Omega_1(\upsilon_1(N_0))$  has p+1 subgroups of order p. Therefore, G has at most p+2 distinct commutator subgroups associated with the maximal subgroups. This contradicts the fact that G has  $p^2+p+1$  maximal subgroups with distinct commutator subgroups when  $(G: \Phi(G)) = p^3$ . Hence,  $(G: \Phi(G)) = p^2$ , which proves part (d).

From (4.1), it now follows that G/A is a Redei group, that is,  $|G'| = p^3 \cdot \text{Since } \upsilon_1(G) = Z(G) \text{ and since } (\varphi(G) \colon \upsilon_1(G)) = p \text{ , then } G' \not Z(G) \cdot \text{However, } G' \text{ is abelian. If } G' \text{ is abelian of type (2,1), then by Proposition 1.4(b), } C_G(G') \text{ is a maximal subgroup of } G \cdot \text{But } \text{then, } Z(G)G' \leq Z(C_G(G')) \text{ and } (C_G(G') \colon Z(G)G') = p \cdot \text{These force } C_G(G') \text{ to be an abelian maximal subgroup, which is a contradiction.}$  Hence G' is elementary abelian of order  $p^3$ .

For the last part of the theorem let M and N be any two maximal subgroups such that M'=N'. Then G/N' has two abelian maximal subgroups M/N' and N/N'. It follows that Z(G/N') has index  $p^2$  in G/N' and that  $Z(G/N')=\Phi(G/N')$ . Therefore, each maximal

subgroup of G/N' is abelian, whence G/N' is itself abelian or Redei. In either case,  $|G'| \le p^2$ , which is a contradiction. Hence, no two maximal subgroups have the same commutator subgroup.

THEOREM 4.2: If  $G \in \mathcal{J} \sim \mathcal{R}$ , if  $|G| \geq p^6$  for p > 3 and if each subgroup of index  $p^2$  is abelian, then G has an abelian maximal subgroup.

Proof: Since  $G \not\in \mathcal{R}$ , G is not itself a Redei group. There exists a Redei subgroup  $R_0 < G$  such that  $R_0 < C_G(Z(R_0))$ . By hypothesis,  $R_0$  is a maximal subgroup of G and  $G = C_G(Z(R_0))$ . It follows, therefore, that  $Z(R_0) = Z(G)$ . By Proposition 3.1, G has at most one abelian maximal subgroup.

Suppose that G has no abelian maximal subgroups, that is,  $G \in \mathcal{J}^*$ . By Theorem 4.1, G has a nonmetacyclic maximal subgroup  $N_0$ ,  $|G'| = p^3 = (G\colon Z(G))$ ,  $Z(G) = \upsilon_1(G)$ ,  $(G\colon \varphi(G)) = p^2$ , and G' is elementary abelian. Now,  $G'\not\in Z(G)$ , but  $(G'\colon G'\cap Z(G)) = (G'Z(G)\colon Z(G)) = (\varphi(G)\colon Z(G)) = p$ . Therefore,  $G_3 = [G,G'] \leq Z(G)$  and the class of G is 3. Thus, G is a regular group by Theorem 1.7(a). By Theorem 1.6,  $|\Omega_1(G)| = (G\colon \upsilon_1(G)) = p^3$ . Hence,  $G' = \Omega_1(G)$ . Also by Theorem 1.6, each element of  $\upsilon_1(G)$  is the  $p^{th}$  power of an element of G.

Let M be any maximal subgroup of G different from N $_0$ . Since M'  $\leq \Phi(M) = Z(M) = Z(G) = \upsilon_1(G)$ , there exists an  $x \in G$  such that M' =  $< x^p >$ .  $|G' < x >| = p^l$ , which by hypothesis, implies that G' < x > is an abelian subgroup of G . Denote  $G/N_0'$  by  $\overline{G}$ . By Theorem 4.1(e),  $\overline{N}_0$  is the only abelian maximal subgroup of  $\overline{G}$ .

Then  $\overline{G'}< x>$  is an abelian subgroup of  $\overline{G}$  of order  $p^3$  and of type (2,1). By Proposition 1.4(b),  $(\overline{G}: \mathcal{C}_{\overline{G}}(\overline{G'}< x>)) \leq p$ . If  $\overline{G'}< x> \leq Z(\overline{G})$ , then  $\overline{G'}< x> \leq \overline{N}_0$ . If, on the other hand,  $\overline{G'}< x> \leq Z(\overline{G})$ , then  $\mathcal{C}_{\overline{G}}(\overline{G'}< x>)$  is a maximal subgroup of G. It follows from  $(\mathcal{C}_{\overline{G}}(\overline{G'}< x>): \overline{G'}Z(\overline{G})) = p$  that  $\mathcal{C}_{\overline{G}}(\overline{G'}< x>)$  is abelian. Thus,  $\overline{G'}< x>$  is always contained in  $\overline{N}_0$ . Consequently,  $x \in N_0$  and  $\overline{N}_0' \leq v_1(N_0)$ . Since M is an arbitrary maximal subgroup different from  $N_0$ , then the commutator subgroup of each nonabelian subgroup different from  $N_0$  is contained in  $v_1(N_0)$ .

Now,  $\upsilon_1(N_0) \triangleleft G$  and  $|G/\upsilon_1(N_0)| = p^4$ .  $G/\upsilon_1(N_0)$  has the nonabelian maximal subgroup  $N_0/\upsilon_1(N_0)$ . However, every other maximal subgroup of  $G/\upsilon_1(N_0)$  is abelian. Hence  $G/\upsilon_1(N_0)$  has precisely one Redei subgroup, which contradicts Theorem 1.1. Therefore, it is impossible for each maximal subgroup to be nonabelian. So,  $G \in \mathcal{J} \supset \mathcal{J}^*$ , that is, G has an abelian maximal subgroup.

The above theorem shows that, for p>3, the minimal non-Redei groups of class  $\mathcal{J}\sim\mathcal{R}$  with order greater than  $p^5$  are actually groups of class  $\mathcal{J}\sim\mathcal{J}^*$ . These groups are included in Theorem 3.1. To obtain a complete classification of the groups in  $\mathcal{J}\sim\mathcal{R}$  with each subgroup of index  $p^2$  abelian, there remains for p>3 only the classification of the groups in  $\mathcal{J}^*\sim\mathcal{R}$  of order less than or equal to  $p^5$ . From Proposition 1.4(c), it follows that only groups of order  $p^5$  must be considered.

THEOREM 4.3: If  $|G| = p^5$  for p > 3 and if  $G \in \mathcal{J}^* \sim \mathcal{R}$ ,

then 
$$G = M_0 < c | c^p^2 = 1 >$$

$$\text{where } M_0 = < a, b | a^p^2 = b^p = 1, d = [a, b] > ,$$

$$[a,c] = b, c^p = d^s \text{ for } s \neq 0 \text{ (mod p) and } [b,c] = a^{rp}c^{tp}$$

$$\text{where } r \text{ is } 1 \text{ or a quadratic nonresidue mod p}.$$

Proof: By hypothesis, G is not Redei. By Proposition 1.4(c), G has at least one abelian subgroup of order  $p^3$ . Thus by Definition 1.4, each subgroup of order  $p^3$  is abelian and each subgroup of order  $p^4$  is Redei. By Theorem 4.1, G has a nonmetacyclic maximal subgroup  $N_0$ , G' is elementary abelian of order  $p^3$ ,  $Z(G) = v_1(G)$ , and  $G/N_0'$  has precisely one abelian maximal subgroup  $N_0/N_0'$ .  $G/N_0' \not\in \mathcal{J}^*$ , but by Proposition 3.2,  $G/N_0' \in \mathcal{J}$ .

Now,  $G_3=[G,G']\leq Z(G)$  which implies that G has class 3. Thus, by Theorem 1.7(a), G is regular. But then, from Theorem 1.6, it follows that  $G'=\Omega_1(G)$ . Also, since  $N_0'\leq Z(N_0)=Z(G)=\upsilon_1(G)$ , there exists an  $x\in G \wedge N_0$  such that  $< x^p>=N_0'$  (Theorem 1.6). Let  $\overline{G}=G/N_0'$ , which has order  $p^H$ . Now  $\overline{x}\in\Omega_1(\overline{G})$ , and  $\overline{G'< x}>\leq \Omega_1(\overline{G})$ . Moreover, since  $\overline{G}$  is regular, then  $|\Omega_1(\overline{G})|=(\overline{G}:\upsilon_1(\overline{G}))=p^3$ . Thus,  $\Omega_1(\overline{G})$  is a maximal subgroup of G. Since  $x\not\in N_0$ ,  $\Omega_1(\overline{G})$  is not abelian. By Lemma 3.4,  $\overline{G}$  is then the group of type (7) in Theorem 3.1. Hence,

 $\overline{G} = \overline{N_0} < \overline{c} | \overline{c}^p = 1 > \text{ where } \overline{N_0} = < \overline{a}, \overline{b} | \overline{a}^p^2 = \overline{b}^p = [\overline{a}, \overline{b}] = 1 >$   $[\overline{a}, \overline{c}] = \overline{b} \text{ and } [\overline{b}, \overline{c}] = \overline{a}^{pp} \text{ for } r = 1 \text{ or a quadratic nonresidue mod p .}$ 

There exist elements a,b from  $N_0$  such that  $aN_0' = \overline{a}$  and  $bN_0' = \overline{b}$ ,

and there exists an element  $c \in G \sim N_0$  such that  $cN_0' = \overline{c}$ . Let d = [a,b]. Then  $c^p \in \langle d|d^p = 1 \rangle = N_0'$ . Since  $c \not\in G' = \Omega_1(G)$ ,  $c^p \neq 1$ . Thus  $c^p = d^s$  where  $s \not\equiv 0 \pmod{p}$ .

Now,  $[a,c] = bd^u$  for some u. Let  $b_1 = bd^u$ . Then  $\overline{b}_1 = \overline{b}$  and  $[a,b_1] = d$ . Since  $[\overline{b},\overline{c}] = \overline{a}^{rp}$ , then  $[b_1,c] = a^{rp}d^t = a^{rp}c^{tp}$  for some t.

The groups described in Theorem 4.3 provide examples which show that class  $\mathcal{J}^*$  properly contains class  $\mathcal{R}$ . The next few theorems lead to the conclusion that for p > 3, these groups are the only groups in  $\mathcal{J}^*$  which are not in  $\mathcal{R}$ .

THEOREM 4.4: If p > 3 and if G  $\epsilon$  3  $^*$   $^*$   $^*$  , then  $|G| \ge p^5$  and each Redei subgroup has order  $p^4$  .

Proof: Since G  $\epsilon$  3 \*  $\circ$  R , there exists an integer  $i \ge 1$  such that each subgroup of index  $p^i$  is a Redei group. By Proposition 1.4(c),  $|G| \ge p^5$ .

Suppose that G has an abelian subgroup of order  $p^4$ . Then, each subgroup of order  $p^4$  is abelian. Since G is not a Redei group, it follows that  $|G| \ge p^6$ . Let  $R_0$  be a Redei subgroup of G.  $|R_0| \ge p^5$ . There exists a subgroup  $R^*$  of G such that  $R_0 < R^*$  and  $(R^*: R) = p$ .  $\mathcal{J}^*$  is a subgroup inherited class, so that by Theorem 4.2,  $R^*$  has an abelian maximal subgroup, which is a contradiction to  $R^*$  non-Redei. Thus, G has no abelian subgroup of order  $p^4$ .

THEOREM 4.5: If G  $\epsilon$  3  $\circ$  R and if  $|G| \ge p^6$  for p > 3, then G  $\not$  3  $\stackrel{*}{\cdot}$  .

Proof: Suppose that  $G \in \mathcal{J}^* \sim \mathcal{R}$ . By Theorem 4.4, each Redei subgroup of G has order  $p^4$ . Each subgroup of G of order  $p^6$  also belongs to  $\mathcal{J}^*$ . In particular, by Theorem 2.2  $G \notin \mathcal{R}$ . Therefore, if a contradiction is reached for a subgroup of order  $p^6$ , then a contradiction is also reached for G. Without loss of generality, it may thus be assumed that  $|G| = p^6$ .

Since each subgroup of order  $p^4$  is Redei, then there is at least one normal Redei subgroup  $R_0$ .  $Z(R_0)$ , as a characteristic subgroup of  $R_0$ , is normal in G;  $G/Z(R_0)$  has order  $p^4$ . By Proposition 1.4(c),  $G/Z(R_0)$  has an abelian maximal subgroup  $M_0/Z(R_0)$ . Thus,  $M_0' \leq Z(R_0)$ . By Theorem 2.2  $M_0 \not\in \mathcal{R}$ . But then, as a consequence of Theorem 4.1,  $M_0' \not\leq Z(G)$ , and a contradiction has been reached. Hence,  $G \not\in \mathcal{J}^*$ .

Corollary 4.5.1: If p > 3 and if G  $\epsilon$   $3^* \circ R$  , then G is a group described by (4.2).

The next theorem characterizes for p=3 the groups in  $3* \sim R$  with each Redei subgroup as a maximal subgroup. This result, along with Theorems 2.2, 3.1, and 4.3, gives a complete classification of the groups in 3, for which each subgroup of index  $p^2$  is an abelian subgroup.

THEOREM 4.6: If p = 3 , if G  $\epsilon$  3 \*  $\sim$   ${\cal R}$  , and if G has a Redei maximal subgroup, then |G| =  $p^5$  and G is one of the following.

(1)  $G = N_0 < c | c^{3^2} = 1 > \text{ where } N_0 = < a,b | a^{3^2} = b^3 = d^3 = 1, d = [a,b] >$  is a nonmetacyclic group, [a,c] = b,  $c^3 = d^3$  for some  $s \not\equiv 0 \pmod{3}$  and  $[b,c] = a^3c^{t3}$  for some t.

(2)  $G = N_0 < c | c^{3^2} = 1 > \text{ where } N_0 = < a, b | a^{3^2} = b^3 = d^3 = 1, d = [a, b] >$  is a nonmetacyclic group,  $[c, a] = b, c^3 = d^3$  for some  $s \not\equiv 0 \pmod{3}$  and  $[c, b] = a^6 c^{t3}$  for some t.

Proof: Since G has a Redei maximal subgroup, then each maximal subgroup of G is Redei. It follows from Proposition 1.4(c), that  $|G| \geq p^5$ . By Theorem 4.1 G' is elementary abelian of order  $p^3$ ,  $Z(G) = \upsilon_1(G)$ , and  $(G: \Phi(G)) = p^2$ . By Theorem 1.7(b), G is not a regular group. Since  $G' \leq \Omega_1(R)$  for each Redei subgroup R, then it follows both that each maximal subgroup of G is nonmetacyclic and that  $G' = \Omega_1(R)$ . Moreover, since each element is in some maximal subgroup, then  $\Omega_1(G) = G'$ .

Let  $N_0$  be a maximal subgroup of G. Denote  $G/N_0'$  by  $\overline{G}$ .  $\overline{G}$  is a nonabelian group, which by Theorem 4.1 has precisely one abelian maximal subgroup  $\overline{N}_0$ . Since  $N_0' \cap \upsilon_1(N_0) = 1$ ,  $\overline{N}_0$  is metacyclic. Let  $\overline{M}$  be any subgroup of  $\overline{G}$  of index  $p^2$  such that  $\overline{M} \not \subset \overline{N}_0$ .  $\overline{M} \subseteq C_{\overline{G}}(\overline{M})$ . If  $\overline{M} < C_{\overline{G}}(\overline{M})$ , then either  $C_{\overline{G}}(\overline{M}) = \overline{G}$  or  $C_{\overline{G}}(\overline{M})$  is a maximal subgroup of  $\overline{G}$ . If  $C_{\overline{G}}(\overline{M}) = \overline{G}$ , then  $\overline{M} \subseteq Z(\overline{G})$  which is a contradiction. If  $C_{\overline{G}}(\overline{M})$  is a maximal subgroup, then  $C_{\overline{G}}(\overline{M})$  is abelian, which is a contradiction. Thus,  $\overline{M}$  is a maximal abelian subgroup of  $\overline{G}$ . This implies that  $C_{\overline{G}}(\overline{R}) \subseteq C_{\overline{G}}(\overline{M}) = \overline{M}$  when  $\overline{M} \subseteq \overline{R}$ . Hence,  $\overline{G} \in \mathcal{J}$ .

By Lemma 3.2,  $3^3 \leq |\Omega_1(\overline{G})| \leq 3^4$ . Now,  $\overline{G}' \leq \Omega_1(\overline{G})$  and  $|G'| = p^2$ . If there exists  $\overline{x} \in \Phi(G) \wedge \overline{G}'$  such that  $\overline{x}^3 = 1$ , then there exists an  $x \in \Phi(G)$  such that  $x^{3^2} = 1$  and  $x^3 \in N_0'$ . But  $x \in N_0$ , which implies that  $N_0' \leq \nu_1(N_0)$ , which is a contradiction. Thus,  $\overline{G}' = \Omega_1(\Phi(\overline{G}))$  and  $\Phi(\overline{G}) < \Phi(\overline{G})\Omega_1(\overline{G})$ .

Case 1:  $|\Omega_1(\overline{G})| = 3^3$ . Then  $\Phi(\overline{G})\Omega_1(\overline{G})$  is maximal in  $\overline{G}$ . Since  $\overline{N}_0$  is metacyclic, Lemma 3.4 implies that  $\overline{G}$  is the group of type (7) in Theorem 3.1, that is,

$$\overline{G} = \overline{N}_0 < \overline{c} | \overline{c}^3 = 1 > \text{where } \overline{N}_0 = < \overline{a}, \overline{b} | \overline{a}^3 = \overline{b}^3 = [\overline{a}, \overline{b}] = 1 > ,$$

$$[\overline{a}, \overline{c}] = \overline{b} \text{ and } [\overline{b}, \overline{c}] = \overline{a}^3.$$

Moreover,  $|G| = 3^5$ .

Let a and b be elements of  $N_0$  such that  $aN_0'=\overline{a}$  and  $bN_0'=\overline{b}$ . Let  $c \in G$  such that  $cN_0'=\overline{c}$ . Since  $G'=\Omega_1(G) \leq N_0$ , then  $c^3 \neq 1$ . Thus,  $N_0'=\langle c^3 \rangle$ . Since  $N_0' \cap \upsilon_1(N_0)=1$ , then  $a^{3^2}=b^3=1$ . Let d=[a,b]. Then  $c^3=d^s$  for some  $s \neq 0 \pmod 3$ . Now  $[a,c]=bd^u$  for some u. Let  $b_1=bd^u$ . Then  $\overline{b}_1=\overline{b}$  and  $[a,b_1]=d$ . Since  $[\overline{b},\overline{c}]=\overline{a}^3$ ,  $[b_1,c]=a^3c^{t3}$  for some t.

 $\underline{\text{Case 2:}} \ \Omega_1(\overline{\textbf{G}}) = 3^{\frac{14}{3}} \ . \ \text{Then, since} \ |\Omega_1(\Phi(\textbf{G}))| = 3^2 \ , \ \text{it follows}$  that  $\Omega_1(\overline{\textbf{G}}) = \overline{\textbf{G}} \ . \ \text{Since} \ \overline{\textbf{N}}_0$  is metacyclic, it follows from Lemma 3.3 that  $\overline{\textbf{G}}$  is the group of type (3) in Theorem 3.1. Thus,

$$\overline{G} = \overline{N}_0 < \overline{c} | \overline{c}^3 = 1 > , \overline{N}_0 = < \overline{a}, \overline{b} | \overline{a}^3 = \overline{b}^3 = [\overline{a}, \overline{b}] = 1 >$$

$$[\overline{c}, \overline{a}] = \overline{b} \text{ and } [\overline{c}, \overline{b}] = \overline{a}^6.$$

Also,  $|G| = 3^5$ .

Let a and b be elements of  $N_0$  such that  $aN_0' = \overline{a}$  and  $bN_0' = \overline{b}$ . Let  $c \in G \sim N_0$  such that  $cN_0' = \overline{c}$ . Then  $c^3 \ne 1$  and  $a3^2 = b^3 = 1$ . Let d = [a,b]. Consequently,  $c^3 = d^s$  for some  $s \ne 0 \pmod 3$ . Since  $[c,a] = bd^u$  for some u, let  $b_1 = bd^u$ . Therefore,  $\overline{b}_1 = b$  and  $[a,b_1] = d$ . Since  $[\overline{c},\overline{b}_1] = \overline{a}^6$ , then  $[c,b_1] = a^6c^{t3}$ 

for some t.

Corollary 4.6.1: If p = 3 and if G  $\epsilon$  3 \*  $\sim$  R , then each Redei subgroup has order 3  $^4$  .

Proof: Let R be a Redei subgroup of G. There exists a subgroup  $R^*$  of G such that  $(R^*:R)=3$ . By Theorem 4.6,  $|R^*|=3^5$ . Hence, R has order  $3^4$ . By the definition of  $3^*$ , each Redei subgroup of G has order  $3^4$ .

A new characterization is obtained for non-Redei metacyclic groups.

THEOREM 4.7: A p-group G is a non-Redei metacyclic group if and only if G has an abelian subgroup of order  $p^4$  and there exists an integer  $i \ge 1$  such that each subgroup of G of index  $p^1$  is Redei.

Proof: Suppose that G is a non-Redei metacyclic group. In view of Corollary 2.4.1, G has an abelian subgroup of order p  $^4$  and G  $\epsilon$  R . Thus G  $\epsilon$  3  $^*$  .

Conversely, suppose that G is non-Redei, that G has an abelian subgroup of order p<sup>4</sup> and that G  $\epsilon$  3 \* . Then  $|G| \ge p^6$ . If p > 3, then by Theorem 4.4, G  $\epsilon$  R . If p = 3, then by Corollary 4.6.1, G  $\epsilon$  R . Thus, by Corollary 2.4.1, G is metacyclic.

#### CHAPTER V

## SUMMARY

Three classes of finite p-groups for p > 2 have been considered in this thesis, each class having been defined by conditions which are placed on the minimal nonabelian subgroups of a group. The role of the Redei subgroups has therefore been emphasized, and the resulting influence of these conditions on the structure of the group has been examined. Several results of this work are summarized here together with directions for future investigation with respect to Redei subgroups of a group.

The study of the two classes R and 3\* has led to two new characterizations for the class of metacyclic groups in terms of Redei subgroups. As mentioned previously, both extraspecial groups and groups with no noncyclic metacyclic characteristic subgroups can be described by means of their Redei subgroups. The question is raised as to whether or not other known classes of nonabelian groups can be defined in terms of their Redei subgroups. Such information could provide alternate areas for the application of these classes.

All nonabelian groups of class 3, for which each subgroup of index  $p^2$  is abelian and for which there is at most one abelian maximal subgroup, have been characterized in Theorems 2.2, 3.1, 4.3, and 4.6. The number of different types shows the extensive nature of class 3. One should recall, however, that the formulation of this class was motivated by a property inherent to the maximal abelian subgroups of a group. It is not surprising, then, that the carry-over of the property

to Redei subgroups would result in a collection of groups whose properties need to be more sharply defined. Classes R and  $\mathcal{J}^*$  are two steps in this general direction.

Theorem 1.1 points out that each non-Redei group has at least two Redei subgroups. It is natural then to ask about the total number of such subgroups and the corresponding conjugate classes of such groups. In this connection, since the metacyclic property is invariant, the question can be raised about the structure of groups for which each Redei subgroup is metacyclic. Information of this sort would extend the result of Blackburn (Theorem 1.10) which classifies groups for which each subgroup is metacyclic. A partial answer to this question is supplied in the present work. Theorems 3.1 and 4.1 indicate that there is only one type of minimal non-Redei group from class  $\mathcal{J} \sim \mathcal{R}$  which has each Redei subgroup as a metacyclic group. This same question could be posed with respect to the nonmetacyclic Redei subgroups.

In both class R and class S, it is seen that there must always exist a normal Redei subgroup. One then wonders if there must always be a normal Redei subgroup. If not, one could look for conditions which would ensure the existence of such normal subgroups. In this regard, there is also raised the question about the structure of a group which possesses characteristic Redei subgroups; two examples would be to consider  $\Phi(G)$  and G' as Redei subgroups.

It should be noticed that Redei subgroups are nilpotent of class 2. Each Redei subgroup, then, is contained in a subgroup maximal with the property of having class 2. An investigation in this area would clarify the position of such subgroups in the theory of finite groups.

In particular, one could consider groups for which the Redei subgroups are maximal with the property of having class 2. The theorems here, dealing with the minimal non-Redei groups, provide some information in this respect.

## BIBLIOGRAPHY

- 1. J. L. Alperin, Large abelian subgroups of p-groups, Trans. Amer. Math. Soc. 117(1965), 10-20.
- 2. H. Bechtell, Theory of Groups, Addison-Wesley, Reading, (1971).
- 3. N. Blackburn, Generalizations of certain elementary theorems on p-groups, Proc. London Math. Soc. 11(1961), 1-22.
- 4. \_\_\_\_\_, On prime power groups in which the derived group has two generators, Proc. Cambridge Philos. Soc. 53(1957), 19-27.
- 5. \_\_\_\_\_, On prime-power groups with two generators, Proc. Cambridge Philos. Soc. 54(1958), 327-337.
- 6. W. Burnside, Theory of Groups, 2nd Edition, 1911, Dover, New York, (1955).
- 7. W. Feit, and J. G. Thompson, Solvability of groups of odd order, Pacific J. Math. 13(1963), 755-1029.
- 8. W. Gaschütz, Gruppen, deren sämtliche Untergruppen Zentralisatoren sind, Arch. Math. 6(1955), 5-8.
- 9. \_\_\_\_\_, Über die Φ-Untergruppen endlicher Gruppen, Math. Z. 58(1953), 160-170.
- 10. D. Gorenstein, Finite Groups, Harper and Row, New York, (1968).
- 11. \_\_\_\_\_, On a theorem of Philip Hall, Pacific J. Math. 19(1966), 77-80.
- 12. M. Hall, The Theory of Groups, Macmillan, New York, (1959).
- 13. P. Hall, A contribution to the theory of groups of prime power orders, Proc. London Math. Soc. 36(1933), 29-95.
- 14. C. Hobby, The Frattini subgroup of a p-group, Pacific J. Math. 10 (1960), 209-212.
- 15. B. Huppert, Endliche Gruppen I, Springer-Verlag, Berlin, (1967).

- 16. B. Huppert, Über das Produkt von paarweise vertauschbaren zyklischen Gruppen, Math. Z. 58(1953), 243-264.
- 17. M. Konvisser, Embedding of abelian subgroups in p-groups, Trans. Amer. Math. Soc. 153(1971), 469-481.
- 18. G. A. Miller, Isomorphisms of a group whose order is a power of a prime, Trans. Amer. Math. Soc. 12(1911), 387-402.
- 19. G. A. Miller, and H. Moreno, Nonabelian groups in which every subgroup is abelian, Trans. Amer. Math. Soc. 4(1903), 398-404.
- 20. L. Rédei, Das schiefe Produkt in der Gruppen, Comment. Math. Helvet. 20(1947), 225-267.
- 21. J. G. Thompson, Nonsolvable finite groups, all of whose local subgroups are solvable, Bull. Amer. Math. Soc. 74(1968), 383-437.